FINANCIAL MONITORING IN THE NEW ASEAN-5 COUNTRIES

Se Hee Lim and Noel G. Reyes

NO. 134

May 2014

ADB WORKING PAPER SERIES ON REGIONAL ECONOMIC INTEGRATION



ASIAN DEVELOPMENT BANK

Financial Monitoring in the New ASEAN-5 Countries

Se Hee Lim^{*} and Noel G. Reyes^{**} No. 134 | May 2014 *Senior Advisor at Korea Center for International Finance and Director at Financial Supervisory Service, Republic of Korea. shlim@fss.or.kr

^{**}Consultant, Office of Regional Economic Integration, 6 ADB Avenue, Asian Development Bank, Mandaluyong City. noelreyes.consultant@adb.org



The ADB Working Paper Series on Regional Economic Integration focuses on topics relating to regional cooperation and integration in the areas of infrastructure and software, trade and investment, money and finance, and regional public goods. The Series is a quick-disseminating, informal publication that seeks to provide information, generate discussion, and elicit comments. Working papers published under this Series may subsequently be published elsewhere.

Disclaimer:

The views expressed in this paper are those of the authors and do not necessarily reflect the views and policies of the Asian Development Bank (ADB) or its Board of Governors or the governments they represent.

ADB does not guarantee the accuracy of the data included in this publication and accepts no responsibility for any consequence of their use.

By making any designation of or reference to a particular territory or geographic area, or by using the term "country" in this document, ADB does not intend to make any judgments as to the legal or other status of any territory or area.

Unless otherwise noted, "\$" refers to US dollars.

© 2014 by Asian Development Bank May 2014 Publication Stock No. WPS146550

Contents	Abstract	i
	ADSITACE	iv
	1. Introduction	1
	2. BCLMV Countries and the Asean Economic Community	3
	3. Diversity of The BCLMV Countries	4
	4. The Key Development Issues	8
	4.1 Financial Regulatory and Supervisory System	8
	4.2 Capital Adequacy and Liquidity Management	15
	4.3 Macroprudential Surveillance	19
	4.4 Disclosure and Transparency	27
	4.5 Capital Flow Management	33
	5. Summary Findings and Recommendations	34
	References	39
	ADB Working Paper Series on Regional Economic Integration	41

Tables

1.	Financial Regulatory Systems of BCLMV Countries	9
2.	Selected Policy Changes in Basel II and III	17
3.	Banking Reforms in BCLMV Countries	18
4.	Summary of Macroprudential Indicators	20
5.	Core vs. Noncore Liabilities (by degree of liquidity)	23
6.	Availability of Data on Core and Noncore Liabilities of the Financial Intermediary Sector in BCLMV Countries	24
7.	Summary: Financial Monitoring of BCLMV Countries	38

Boxes

1.	Organisation for Economic Co-operation and Development	
	Principle on Disclosure and Transparency	29
2.	Financial Sector Development Priorities in the BCLMV Countries	36

Abstract

This paper examines the issues surrounding the implementation of global regulatory reforms spearheaded by the G20 and mainly under the aegis of the Financial Stability Board (FSB)—in Brunei Darussalam, Cambodia, the Lao People's Democratic Republic (Lao PDR), Myanmar, and Viet Nam (BCLMV). These countries are the five newest members of the Association of Southeast Asian Nations (ASEAN). As such, there has been little consideration of the impact of global regulatory reforms on these countries. This paper contributes to the literature by providing an analysis of the capacity of the BCLMV countries to implement necessary financial regulatory reforms. Further, this analysis supplements ongoing efforts to establish the building blocks for the ASEAN Economic Community (AEC), which is scheduled to be implemented by 2015. Toward this end, the paper addresses five key development issues in the BCLMV countries: (i) financial regulatory and supervisory systems, (ii) compliance with capital adequacy and liquidity management guidelines under the Basel reforms, (iii) macroprudential surveillance systems, (iv) disclosure and transparency, and (v) capital flow management.

Keywords: financial regulatory systems; capital adequacy and liquidity management; Basel reforms; macroprudential surveillance; disclosure and transparency; capital flow; noncore liabilities; BCLMV and ASEAN-5 countries; and ASEAN Economic Community

JEL Classification: G28; K22; and O16

1. Introduction

Five years after its onset, the adverse effects of the global financial crisis (GFC) continue to be felt in the developed economies of the United States (US) and the eurozone. At the same time, the developing economies of Asia have remained relatively unscathed from the entire episode. The US economy's recovery remains anaemic and its unemployment rate, while down from its peak, remains stubbornly high even after the US government and Federal Reserve have instituted extraordinary measures and pumped hundreds of billions of dollars into the economy to stabilize financial markets and boost economic activity. In the eurozone, member economies have suffered recurrent bouts of recession characterized by sovereign debt defaults and steep credit downgrades, bank insolvencies, and record rates of joblessness.

The debilitating effects of the GFC and the overriding objective of preventing another similar crisis from occurring provided the impetus for the major economies to introduce a number of extensive regulatory reforms. These global reform measures—spearheaded by the Group of 20 Nations (G20), Financial Stability Board (FSB), International Monetary Fund (IMF), and Bank of International Settlements (BIS)—seek to bolster the resilience and stability of financial systems by addressing the vulnerabilities exposed by the crisis.

In addition, a new global financial architecture emerged in the wake of the GFC, with the FSB at its apex. Established after the G20 London summit in April 2009 as a successor to the Financial Stability Forum (FSF), the FSB coordinates the work of national financial authorities and international standard setting bodies. It also develops and promotes "the implementation of effective regulatory, supervisory, and other financial sector policies." Its members include all of the major economies that comprise the G20, former FSF members, and the European Commission.

The FSB coordinates proposed global reform measures, which may be broadly categorized into two main areas of interest.

The first area concerns safety and soundness standards for financial firms, including bank capital and liquidity regulations, international accounting standards, and the resolution of financial firms. These standards cover the Basel III capital framework for banks, the overhaul of the international financial reporting standards under the auspices of the International Accounting Standards Board (IASB) and the US Financial Accounting Standards Board (FASB), and the mandatory establishment of the bank recovery and resolution regime in G20 member countries.

The second area of interest for global reform is the regulation of high-risk activities, including over-the-counter (OTC) derivatives trading, shadow banking, and high-frequency trading. These are to be covered by proposals for, respectively, central counterparty clearing for OTC derivatives, the strengthening of shadow banking oversight with focus on hedge fund activities, and the curtailment or ban of high-frequency trading.

Evident from this discussion is that policymakers in developed economies—the starting point of the crisis and home to most of the world's largest financial institutions—have been setting

the agenda for global regulatory reform. These proposed reforms understandably reflect their concerns and largely tackle the weaknesses and vulnerabilities of the developed economy financial systems that helped trigger the crisis. Yet, this focus on developed economy sources of weakness has left the proposed reform measures open to criticism that they fail to adequately address the distinctive risks and concerns of Asia's emerging economies.

In contrast to the developed economies, for instance, emerging economies mainly suffer from instability brought about by external shocks that upset trade flows and reverse capital flows, pummelling domestic currencies and the prices of local assets in the process and thereby stressing the economy. Furthermore, financial sector development in emerging economies remains in a nascent stage, in which the sector relies mainly on banks for financial intermediation. Placing a regulatory burden on Asia's financial sector, either prematurely or heavy-handedly, could constrain the growth of emerging economies at a crucial stage of their development.

There are those who argue that the reform agenda should be enlarged to accommodate the Asian view. Others argue that implementation of the proposed reforms, even as they relate largely to advanced economies and their sophisticated markets, will still benefit Asia's emerging economies and embryonic markets since these reforms serve the purpose of enhancing financial stability, preventing financial crises, and providing guidance to Asian policymakers with respect to best practices. The last point is especially relevant to the regional integration initiatives of the Association of Southeast Asian Nations (ASEAN), which require the harmonization of regulatory regimes across the region in order to avoid the risk of regulatory arbitrage. In addition, implementation of these reforms would help attract foreign investors who seek the presence of these measures to mitigate investment risks. The ultimate aim would be wean the region from relying too much on capital inflows by developing the region's capital markets and financial systems. However, the patent reality is that the region will remain reliant on these foreign inflows for the time being.

With these contentious points in mind, this paper seeks to study the impact of global regulatory reforms on the five newest members of ASEAN: Brunei Darussalam, Cambodia, the Lao People's Democratic Republic (Lao PDR), Myanmar, and Viet Nam (known collectively as the BCLMV countries). In assessing the impact of such reforms, this paper examines a number of key development issues in the BCLMV countries, including (i) the resilience of financial regulatory and supervisory systems, (ii) compliance with capital adequacy requirements and liquidity management guidelines under the Basel reforms, (iii) macroprudential surveillance systems, (iv) transparency and disclosure, and (v) capital flow management.

Face-to-face interviews were conducted with officials from the five countries' monetary authorities, market regulators, and stock exchanges. Interviewees were emailed a set of questions in advance along with a survey questionnaire based on the principles of corporate governance as determined by the Organisation for Economic Co-operation and Development (OECD) and the BIS.

Most research on financial development and reforms has focused on the more developed or larger ASEAN economies and has generally neglected the five newer members.¹ This study hopes to break new ground by focusing on the BCLMV countries and the status of their financial development and reform programs.

The relevance of the proposed global financial reform measures is particularly acute for these five developing economies, one of which, Brunei Darussalam, is also an oil-exporting country. Their financial sectors lag the rest of the region in terms of development, with banks functioning as the main financial intermediaries that generally offer only the traditional products and services of deposit-taking and lending. The looming global banking reforms, on the other hand, pertain to more advanced financial systems.

In addition to the relevance of global financial reform, another factor considered by this study is the impending economic integration of the ASEAN region by 2015.

The remainder of this paper is structured as follows. Section 2 discusses issues related to BCLMV countries and the ASEAN Economic Community (AEC). Section 3 analyzes issues related to the diversity of the BCLMV countries. Section 4 deals with various key development issues and indicators. Finally, section 5 summarizes and offers some recommendations.

2. BCLMV Countries and the ASEAN Economic Community

An overarching concern is the possibility that the BCLMV countries' financial development will continue to lag the rest of the region, which is currently gearing up for the accelerated establishment of the AEC by 2015. Upon its official launch, the 10-country integrated region will cover 4.46 million square kilometers, with a combined population of 600 million and a combined nominal gross domestic product (GDP) of more than \$2 trillion.

The official launch of the AEC has been moved forward from its original date. During the 13th ASEAN Summit on 20 November 2007 in Singapore, ASEAN leaders agreed to move from 2020 to 2015 the establishment of the AEC in the hope of accelerating the economic integration of the region. The ASEAN leaders adopted the AEC Blueprint in 2008 to serve as the master plan for the formation of the AEC.

The community's areas of cooperation include human resource development and capacity building, recognition of professional qualifications, closer consultation on macroeconomic and financial policies, trade financing measures, enhanced infrastructure and communication connectivity, development of electronic transactions through e-ASEAN, integrated industries across the region to promote regional sourcing, and enhanced private sector involvement in the establishment of the AEC.

ASEAN was formed on 8 August 1967 by Indonesia, Malaysia, the Philippines, Singapore, and Thailand. Since then, membership has expanded to include Brunei Darussalam, Myanmar, Cambodia, the Lao PDR, and Viet Nam.

The AEC will transform ASEAN into a region "with free movement of goods, services, investment, skilled labor, and freer flow of capital" (AEC Blueprint 2008). With regard to the "freer flow of capital," the AEC Blueprint calls for strengthening ASEAN capital market development and integration, and allowing greater capital mobility in the region.

To achieve the goal of capital market integration, the planned policy actions include greater harmonization of standards for debt securities, disclosure requirements, and distribution rules; cross-recognition of market professionals; greater flexibility in language and the governing legal requirements for securities issuance; enhanced withholding tax structures to promote the broadening of the investor base in ASEAN debt issuance; and the establishment of exchange and debt market linkages.

To allow for greater capital mobility, the blueprint calls for removal or relaxation of restrictions on current account transactions to facilitate the flow of payments and transfers for current account transactions. With regard to capital flows, the blueprint supports foreign direct investment (FDI) and initiatives to promote capital market development.

Not all 10 of ASEAN's members will be ready to implement these policy actions by 2015. The BCLMV countries, in particular, need to catch up with other ASEAN members in terms of financial sector and capital market development, as well as the adoption of global reform measures. ASEAN leaders recognize the different levels of development among its members, which "require[s] some flexibility as ASEAN moves towards a more integrated and interdependent future" (AEC Blueprint 2008). With this in mind, the AEC Blueprint intends to "address the development divide and accelerate integration of Cambodia, the Lao PDR, Myanmar, and Viet Nam (CLMV) through the Initiative for ASEAN Integration and other regional initiatives."

The Initiative for ASEAN Integration, launched in November 2000, will be enhanced to serve as the platform for identifying and implementing technical assistance and capacity building programs for both the public and private sectors in ASEAN member countries, particularly in CLMV countries and among other sub-regional arrangements.

3. Diversity of the BCLMV Countries

Any comparative discussion of the need for implementing economic reforms in the BCLMV countries must be cognizant of the extensive differences and diversity in terms of income, stage of development, political and economic structures, culture, and demographics, among other characteristics (Table 1). These five countries' political frameworks, for example, range from an absolute monarchy to communist states to a parliamentary system. Their income levels range from as low as about \$120 to more than \$30,000 per capita. A number of commonalities also exist, such as the underdevelopment of the financial sector and undercapitalization of banks, less-developed to undeveloped capital markets, weak legal and regulatory enforcement, and wide gaps in human capital capacities.

The AEC Blueprint clearly recognizes these differences and makes the following concession: "Liberalization measures [in] the financial services sector should allow members to ensure orderly financial sector development and maintenance of financial and socio-economic stability." Specifically, it sets guidelines for member countries in pacing their liberalization measures, providing the "ASEAN minus X formula, where countries that are ready to liberalize can proceed first and be joined by others later." It added that the "process of liberalization should take place with due respect for national policy objectives and the level of economic and financial sector development of the individual members."

In the same vein, a joint Asian Development Bank (ADB)–ASEAN publication in 2013 proposed an alternative approach that would "allow financial integration in different member states to proceed at different speeds in order to build regional financial market infrastructure and harmonize institutions, market practices, and policies, and thus lay the foundation for regional financial integration." The study sets two different frameworks for capital account liberalization and ASEAN-wide financial integration: one for the five original members (ASEAN-5) and another for the five newcomers (BCLMV).

Brunei Darussalam

The oil and gas sector dominates Brunei Darussalam's economy, with crude oil and natural gas production accounting for over half of GDP and more than 90% of exports, while employing less than 3% of the country's workforce. Revenues from this sector also comprise most of the country's GDP per capita of about \$31,000—one of the highest in Asia and the second-highest among ASEAN nations—and finances an extensive social welfare program under which the government provides for all medical services, offers free education through the university level, and subsidizes rice and housing for all of its citizens. The government is also the country's largest employer. In early 2013, there were proposals at the Legislative Council to shorten the probationary period of service from 5 years to 3 years for civil servants being paid on a daily basis to become paid on a monthly basis. The government employs nearly 50,000 regular employees (excluding probationary workers), in comparison with private sector employment of about 121,000 in all industries. Another major recent proposal was the introduction of a minimum wage law.

Brunei Darussalam operates a currency board system with the Brunei Dollar (B\$) pegged to the Singapore Dollar (S\$) at the rate of 1:1. The country has a low tariff regime and no capital gains or personal income taxes. Under its Investment Incentives Order 2001, prospective investors enjoy a wide range of incentives, including up to 20 years exemption from corporate taxes; exemption from import duties on raw material, machinery, equipment, component parts, accessories and building structures; and adjustment of capital allowance and losses.

On the other hand, the rest of the domestic economy lags in terms of development and progress. Economic growth has also been a laggard, averaging around 0.2% over the past 5 years, which is well below the 1.8% average rate at which Brunei Darussalam's population is growing. The country's GDP growth took a hit during the GFC as the resulting drop in world crude prices and slump in oil demand pushed the economy into a long recession.

Excessive reliance on the oil and gas sector as well as the steady depletion of these natural resource reserves has become a major concern for Brunei Darussalam's leaders. *The Brunei Darussalam Long-Term Development Plan: Wawasan Brunei 2035* articulated the key development issue facing the country: "Although oil and gas resources have contributed much to the nation's prosperity, economic growth has, on the whole, not kept pace with population growth. The public sector that is the main employer of the majority of the citizens and residents can no longer adequately absorb the growing numbers of young people wishing to enter the work force each year. There is a widening gap between the expectations and capabilities of the nation's youth and the employment opportunities currently being created."

Cambodia

Cambodia remains largely a dollarized economy, a system that came about as a result of historical circumstances rather than a conscious decision by the country's leaders.² The use of US dollars in Cambodia had its genesis during the Khmer Rouge era (1975–1979) when all barter, private commercial activity, private ownership, means of exchange, and stores of value were prohibited and punishable by death. During this period, Cambodia was without a monetary system and without money. In 1980, the riel was reintroduced but it did not receive wide acceptance by the public, which preferred other stores of value and means of payment such as the US dollar, gold, and even rice.

GDP growth has averaged more than 6% per year from 2010 to 2012, driven largely by garment manufacturing, construction, agriculture, and tourism. The garment industry employs more than 335,000 people and accounts for more than 75% of Cambodia's total exports. Oil deposits were discovered in 2005 but remain unexploited. Investors are also looking to exploit mining opportunities that include deposits of bauxite, gold, iron, and gems. Lastly, the tourism industry remains a growth sector, with foreign visitor arrivals surpassing 2 million per year.

Cambodia, however, remains one of the poorest countries in Asia, with GDP per capita below \$900 and with about 4 million people living on less than \$1.25 per day and 37% of Cambodian children under the age of 5 years old suffering from chronic malnutrition. More than 50% of the population is less than 25 years old. The population lacks education and productive skills, particularly in the impoverished countryside, which also lacks basic infrastructure.

More than half of the government budget comes from bilateral and multilateral donors, which have tied foreign aid to government passage of economic reform measures.

The country's economy was hit by the GFC, with its critically important garment industry suffering a 23% drop in exports to the US and Europe. GDP growth slumped to 0.1% in 2009 before recovering to more than 7% growth in 2011. It suffered no direct impact from the crisis in its financial sector and markets, however. The main lessons from the GFC, as far as the Government of Cambodia is concerned, include the need for less intervention in the markets and the need to bestow more power with the securities regulator.

² A detailed history of Cambodia's currency may be found in G. Capannelli and J. Menon (2010).

Lao People's Democratic Republic

Lao PDR is in a transition period where the government is in the process of modernizing the economy. This endeavor is being supported by the multilaterals, with experts reckoning the process might take 10–15 years.

Since the Communist party started decentralizing control and encouraging private enterprise, the country's GDP, while coming from a very low base, has grown an average of 6% from 1998 to 2008, and more than 7% from 2008 to 2012. The Lao PDR joined ASEAN in 1997 and the World Trade Organization (WTO) in 2013, and gained Normal Trade Relations status with the US in 2004.

Its economy continues to rely on subsistence agriculture, dominated by rice cultivation in lowland areas, which accounts for about 30% of GDP and 75% of total employment. Its landlocked economy also relies on investment from and trade with its neighbors: the People's Republic of China (PRC), Thailand, and Viet Nam. The country's water resources and mountainous terrain enable it to produce and export large quantities of hydroelectric energy to Thailand and Viet Nam. Tourism is the fastest-growing industry in the country.

Experts note complications in the Lao PDR's political economy given that the country is governed under a one-party system. The politburo, comprising the leadership of the Peoples' Revolutionary Party, sets policies, which the government bureaucracy then implements. The current hierarchical system also places the Ministry of Planning above the Ministry of Finance, which in turn supervises the Bank of Lao PDR (BOL).

Myanmar

Myanmar remains a country in profound transition, in terms of its political economy and economic structure. The country embarked on a path of political and economic reforms in 2011, paving the way for developing its large potential. According to an ADB study in 2012, Myanmar could follow Asia's fast-growing economies and expand at 7%–8% per year, become a middle-income nation, and triple its per capita income by 2030 if it can surmount substantial development challenges by further implementing across-the-board reforms.

In August 2012, ADB stepped up its presence in Myanmar by assigning more staff to extended missions in the country. On 28 January 2013, ADB resumed its loan operations in Myanmar. The \$512 million loan, the first from ADB in almost 30 years, was made possible through bridge financing provided to the government by the Japan Bank for International Cooperation (JBIC). The loan will be used to finalize arrears clearance and sustain government efforts to revamp the national budget process and modernize tax administration. It will also support trade policy reforms and capacity development, and improve the investment climate and facilitate small and medium-sized enterprise (SME) development.

Myanmar's development and poverty reduction challenges remain substantial and require considerable external assistance. At the same time, Myanmar faces enormous development challenges following a long period without comprehensive assistance from the international

community. Strong and timely financial commitments are crucial to creating a foundation for inclusive economic development and maintaining the reform momentum in Myanmar.

Much still needs to be done, however, as the domestic economy continues to suffer from pervasive government controls, inefficient economic policies, corruption, and rural poverty. About 32% of the population lives in poverty and Myanmar is the poorest country in Southeast Asia, with GDP per capita of just over \$120.

Meanwhile, political parties have begun gearing up for the next round of general elections in 2015, potentially ushering in renewed political uncertainties.

Viet Nam

Among the BCLMV countries, Viet Nam has the most developed financial sector, making it the model of transitional economic development among its less-developed neighbors. Since its *doi moi* (renovation) policy was launched in 1986, the country has been transitioning from a centrally planned economy to a socialist-oriented market economy.

Agriculture's share of economic output shrank from about 25% in 2000 to less than 22% in 2012, while industry's share increased from 36% to nearly 41% in the same period. State-owned enterprises (SOEs) account for roughly 40% of GDP.

The GFC hurt Viet Nam's export-oriented economy, with GDP in 2009–12 growing less per year than the 7% per annum average achieved in the prior decade. Between 2008 and 2011, Viet Nam's managed currency, the dong, was devalued by more than 20%.

Government economic policy in recent years has swung back and forth from a growthoriented strategy to one that focuses on fostering macroeconomic stability. At the start of 2011, the government veered away from growth stimulation, which had stoked inflation to doubledigit rates, and instead tightened monetary and fiscal control to stabilize the economy. The following year the government unveiled a broad "three pillar" economic reform program involving the restructuring of public investment, SOEs, and the banking sector. Viet Nam's economy continues to face challenges from an undercapitalized banking sector. Nonperforming loans (NPLs) weigh heavily on banks and businesses. In September 2012, the official bad debt ratio climbed to 8.8%, though some financial analysts believe it could be as high as 15%.

4. The Key Development Issues

4.1 Financial Regulatory and Supervisory System

The financial system consists of institutions, markets, laws, regulations, and practices through which securities are traded, interest rates determined, and financial services produced and delivered. Markets—consisting of a market for short-term funds (money market) and another for long-term funds (capital market)—channel loanable and investible funds to those who

demand them and are willing to pay the cost. Institutions, on the other hand, consist of banks, insurers, finance and investment companies, and other financial intermediaries. Atop these markets and institutions are government regulators, whose main functions include preserving the safety of the public's funds, promoting public confidence in the financial system, and supporting the stability of markets and the economy, among others. The standard structure for government regulation calls for three independent agencies to oversee the banking system, the capital markets, and the insurance sector. A few developed economies have merged two or more of these functions under one agency. In the underdeveloped financial systems of the BCLMV countries, however, one or two of these essential agencies are either non-existent or only at their nascent stages as new laws are being enacted and new rules and regulations are being promulgated to create them as quickly as possible.

The table below summarizes the financial regulatory and supervisory systems of the BCLMV countries.

	Regulator / Supervisor	Remarks
Brunei Darussalam	Autoriti Monetari Brunei Darussalam (AMBD)	Established 1 January 2011 under AMBD Order, 2010. AMBD administers and enforces: Banking Order, 2006; Islamic Banking Order, 2008; International Banking Order, 2000; Finance Companies Act, Chapter 89; Hire Purchase Order, 2006; Pawnbrokers Order, 2002; and Money-Changing and Remittance Businesses Act, Chapter 174.
	No separate securities regulator	1 1
Cambodia	National Bank of Cambodia (NBC)	Established in 1954 Manages monetary and exchange policies, regulates banks and financial institutions, and manages the national currency, the riel.
	Securities and Exchange Commission of Cambodia (SECC)	Inaugurated in April 2009 under Issuance and Trading of Non- Government Securities (Preah Reach Kram No NS/RKM/ 1007/028) Regulator managing, supervising, and developing the securities sector in Cambodia. Minister of Economy and Finance serves as the SECC Chairman.

Table 1: Financial Regulatory Systems of BCLMV Countries

10 | Working Paper Series on Regional Economic Integration No. 134

Table 1: Continued

	Regulator / Supervisor	Remarks
Lao People's Democratic Republic	Bank of Lao P.D.R. (BOL)	Officially founded in 1988, the bank's charter was adopted by the law No. 05/NA, dated 14 October 1995, with the bank equivalent to a ministry in rank.
	Lao Securities and Exchange Commission Office (SECO)	Supervises the Lao Securities Exchange (LSX) SECO and LSX remain BOL units
Myanmar	Central Bank of Myanmar	Created under the Central Bank of Myanmar Law (1990), which makes it responsible for financial stability and supervision of the financial sector in Myanmar. Currently under the Ministry of Finance and Revenue New law to grant CBM independence under parliament debate
	No separate securities regulator	In the pipeline are new laws on financial institutions, foreign exchange management, and securities exchange.
Viet Nam	State Bank of Vietnam (SBV)	Had its beginning in 1951. Law on the State Bank of Vietnam and the Law on Credit institutions, effective from January 1998.
	State Securities Commission (SSC)	Established in 1996, oversees and regulates securities trading on the country's two official exchanges Remains officially part of the MOF

Source: Authors' compilation.

Brunei Darussalam. Autoriti Monetari Brunei Darussalam (AMBD)—established on 1 January 2011 with the introduction of AMBD Order, 2010—is a corporate body acting as the central bank of Brunei Darussalam and undertaking core functions, including the formulation and implementation of monetary policy, the regulation and supervision of financial institutions, and currency management. The adoption of the currency board system, however, negated the monetary policy functions of the AMBD.

In its supervision and regulation of banks and finance companies, AMBD administers and enforces the following laws: Banking Order, 2006; Islamic Banking Order, 2008; International Banking Order, 2000; Finance Companies Act, Chapter 89; Hire Purchase Order, 2006; Pawnbrokers Order, 2002; and Money-Changing and Remittance Businesses Act, Chapter 174. As part of its functions, in March 2013, AMBD imposed floors and ceilings on banks' deposit and lending rates (Autoriti Monetari Brunei Darussalam 2013). Banks cannot charge more than 4.5% on residential property loans, between 5.0% and 5.25% on consumer loans (excluding credit cards), 6.0% on corporate loans, 7.5% on salary and pension loans, and between 7.725% and 7.75% on new vehicle loans. The minimum rates on savings were set at 0.15% and between 0.20% and 0.75% for 1- and 12-month time deposits, respectively. The monetary authority reckoned that the market mechanism failed to reflect the costs of credit and funds because of the small number of players in the banking industry. Operating in the country are a total of nine banks, including the trust fund, six foreign banks (three regional and three international banks), two Islamic banks, and one private indigenous bank.

The banks' gross NPL ratio had previously averaged about 9% of total loans, with a higher ratio of 12.3% for domestic banks. More recently, AMBD officials disclosed that the banking industry's gross NPLs had exceeded 10%. After provisioning, net NPLs were placed below 2%, as AMBD mandated banks to fully write off their "legacy loans." NPLs primarily consist of personal loans (including credit cards), which account for roughly 40% of total banking industry loans.

The current priorities of the central bank are to enhance the national payments system, set up a trade information bureau, and establish a deposit insurance scheme under the Ministry of Finance.

Cambodia. The National Bank of Cambodia (NBC), established in 1954, is the country's central bank and, as such, manages monetary and exchange policies; regulates banks and financial institutions; and manages the national currency, the riel.

In 2009, NBC responded to the GFC by raising reserve requirements to 16% from 8% before the crisis, but it has since reduced this to 12.5%. Liquidity is maintained at 50% of net worth. Local banks have no access to investments; instead, the central bank "invests prudentially for them" using international reserves of \$3.4 billion.

To coordinate the functions of the fiscal and monetary authorities, the heads of the Ministry of Economy and Finance (MEF) and NBC sit as chair and vice-chair, respectively, of the Economic and Financial Policy Committee, which meets on an ad hoc basis. The Supreme National Economic Council, a think tank for the government under the office of the prime minister, aids the government in formulating public policies concerning macroeconomic and sectoral issues.

NBC's priority capacity building efforts focus on the following: ASEAN integration, mainly with respect to the 2015 deadline for implementation of the AEC; development of capital account monitoring; development of effective monetary policy (open market operations and benchmark interest rates); banking supervision and regulation (onsite and offsite); strengthened payments system; and a modern information system.

The Securities and Exchange Commission of Cambodia (SECC) was inaugurated in April 2009 under the law on the Issuance and Trading of Non-Government Securities (Preah Reach Kram

No NS/RKM/ 1007/028) as the regulator managing, supervising, and developing the securities sector in Cambodia. The MEF serves as the SECC Chairman.

The SECC, together with the MEF and NBC, comprise the crisis preparedness management team created in 2012. This team is now undertaking research on crisis response activities. Further, a private sector-government forum meets every 6 months.

The development priorities of the SECC include attracting more companies to list and preparing for financial integration in the region. It also seeks to build regulatory capacity in the areas of initial public offering (IPO) evaluation, processing, and procedure; listing promotion; corporate governance; market surveillance; and market investigation.

The Lao People's Democratic Republic. Under the supervision of the Bank of Lao PDR (BOL) are 31 banks, of which 16 are foreign and three are subsidiaries of foreign banks. The regulator conducts offsite supervision and monitoring on a daily basis and comprehensive onsite evaluations annually, coordinating these activities with one another. Founded in 1988, the bank's charter was adopted by law No. 05/NA, dated 14 October 1995, with the bank being given status equivalent to a ministry.

Banks in Lao PDR on average have NPLs of 2%, which is below the 3% threshold level monitored by the central bank given the country's present stage of development. The BOL, furthermore, does not set limits on bank lending to specific sectors, but allows a single borrower limit of 25%. The BOL sets no loan-to-value (LTV) ratios and has no debt-to-income (DTI) rules.

There are some doubts over the official NPL figure of only 2% given that loan growth averages around 20% per annum. Furthermore, with regard to infrastructure lending, foreign and local contractors borrow from banks with an implicit sovereign guarantee.

Bank lending rates currently average 15% in US dollar terms, while rates on 1-year time deposits average 8%.

Although the country's latest medium-term development plan lacks a financial sector plan, government officials have noted that the banking sector is subsumed under the plan's "monetary policy and financial stability" section.

Priorities for the finance, banking, and capital market sectors under the Seventh Five-Year National Socio-Economic Development Plan (2011–15) include financial integration with the region, development of the securities market, and development of the capital market. The plan's goals include raising the number of publicly listed companies to at least 10 by 2015, expanding the number of products traded in the stock market, and broadening the investor base (individuals and institutions).

The Lao Securities and Exchange Commission Office (SECO) and the Lao Securities Exchange (LSX) remain under BOL authority, even as government officials understand that the securities regulator should be independent. The PRC, Thailand, and Viet Nam act as models

for the SECO, which currently has over 40 people headed by an official of director general rank. It has partnered with the Thai and Vietnamese securities and exchange commissions, and forged an agreement with the SEC of Malaysia to train its staff on drafting a strategic plan for the development of the capital market.

In case of a financial crisis, the focal point would be the Ministry of Finance (MOF), which maintains a network of communications with the line ministries and the BOL. Every month, economic data is provided to the MOF, whose Fiscal Policy Department then analyzes the data. In addition, the deputy prime minister heads a high-level committee that includes the BOL Governor, the MOF, and the Prime Minister's Office.

The MOF, furthermore, administers the country's SOEs and has drafted a government decree on the transferring ownership of these SOEs to the public. The MOF also provided listing candidates during establishment of the SECO.

Myanmar. Following a series of political and economic reforms that led to a significant opening of the country and the lifting of international sanctions against it, Myanmar is now rushing to reform its financial system. It is currently rewriting its banking and financial laws under the auspices of the IMF and the World Bank.

A new law granting independence to the Central Bank of Myanmar (CBM), currently operated by the Ministry of Finance,³ will be debated in Parliament. The CBM was created under the Central Bank of Myanmar Law (1990), which makes it responsible for financial stability and supervision of the country's financial sector. Also in the pipeline are new laws on financial institutions, foreign exchange management, and a securities exchange.

CBM currently oversees four SOE banks, 19 private banks, and 28 representative offices of foreign banks. It uses two main approaches for supervisory and regulation, and monitoring: onsite examination and off-site monitoring. On-site examination uses the CAMEL framework,⁴ while off-site monitoring operations are based on the weekly, monthly, quarterly, and annual reports submitted by the banks. Banks are also required to submit audited financial statements once a year. CBM officials, however, admit that accounting and disclosure requirements remain weak. Furthermore, there is a need to upgrade the system so that it provides standard prudential regulation.

The adverse side effects of the financial sector's lack of development may be seen in the lack of convenience in banking transactions, high cost of intermediation, and basic payments system. Households use savings passbooks as "current deposits" mainly for transactions, while corporations use demand deposits. Time deposits, recently introduced into the market, remain unfamiliar to depositors. On the other hand, the country already has deposit insurance through state-owned Myanmar Insurance.

³ Formerly known as Ministry of Finance and Revenue until July 2013.

⁴ This refers to the international bank-rating system where bank supervisory authorities rate institutions according to the following factors: capital adequacy, asset quality, management quality, earnings, and liquidity.

On top of the new laws being drafted, officials note that the country's economic culture needs to change as well, with institutional capacity building being the most important objective at this stage, especially since most of the people with institutional knowledge are retired or deceased.

Viet Nam. The State Bank of Viet Nam (SBV) supervises about 45 domestic banks in an industry that is dominated by five state-owned banks. Agriculture Bank is the largest in terms of total resources, followed by BIDV Bank, Vietin Bank, and Vietcom Bank. State-owned banks must follow the same regulations and abide by the same corporate governance rules as private banks.

The SBV has its beginnings in 1951 when President Ho Chi Minh signed decree 15/SL to establish the National Bank of Viet Nam, later renamed State Bank and merged with the National Bank of Viet Nam upon the country's reunification. The Law on the State Bank of Viet Nam and the Law on Credit Institutions were passed in December 1997 by the 10th National Assembly of the Socialist Republic of Viet Nam.

There are 100 foreign banks operating in Viet Nam, including branches, subsidiaries, and representative offices. The latter are required to maintain charter capital of at least \$15 million. In terms of foreign ownership, there is a 20% limit for a single investor and a 30% limit for total foreign ownership. There are five listed banks. In terms of disclosure, listed banks follow stock exchange rules, which include the submission and publication of annual audited financial statements and semi-annual statements.

The SBV does not have an overarching LTV policy, rather each bank has its own LTV policy. The SBV is now considering LTV regulations even as LTV ratios average a low of 60%.

Banks' NPLs have grown as a result of the recent macroeconomic imbalances that include high inflation and slower growth. As a result, "enterprises have faced difficulties," with NPLs reaching a high of 9% of total loans in 2012.⁵ In response, the government issued a master plan to restructure SOEs, the economy, and the banking system, including boosting economic growth and quickly resolving the real estate and NPL issue. The SBV also issued a directive for banks to set aside funds for socialized housing.

Prime Minister Nguyen Tan Dung approved the formation of an asset management company, effective 9 July 2013, to deal with the banking sectors' bad debts problem together with the

⁵ Private estimates put the banking sector's NPLs at a much higher rate of 15%. In 2012, the SBV conducted comprehensive supervision of some weak banks, which revealed that the small banks had bad loan ratios of up to 60%. The SBV has since then upgraded provisioning requirements for banks. Another source of complication is cross-shareholdings among banks. Also, small banks lend only to SOEs, making them de facto financial holding companies for these SOEs. Adding to the banking system's woes is that the real estate market has been either frozen or suffering from price declines for the past 2 years. CBRE, a private realty consultant, recorded declining trends in office rentals for the past 2 years due to oversupply and sees this trend continuing in the near-term. A lot of these office projects are funded by SOEs. Furthermore, borrowers usually use real estate as collateral. As a result, an estimated 60%–70% of NPLs are real-estate linked. Also under a cloud of doubt are the accounting standards being used. Viet Nam accounting standards (VAS), with the MOF acting as the national accounting body, still need to be aligned with international accounting standards (IAS). Further, the State Auditor is seen as problematic and weak.

existing Debt and Asset Trading Company (DATC). The Viet Nam Asset Management Corporation (VAMC), a wholly state-owned company managed by the SBV, will have initial capital of D500 billion (\$24 million). According to the SBV Governor Nguyen Van Binhand, VAMC will be tasked to resolve between D40 trillion and D70 trillion of NPLs this year out of total NPLs estimated at about D100 trillion by the chairman of the National Financial Supervisory Commission. The ultimate aim is to accelerate the country's banking restructuring process, restore the flow of credit to the business sector, and rejuvenate the country's economic growth. Credit growth has slowed as a direct consequence of the high rate of bad debts as banks have become more conservative in response.

Also, as part of the master plan, the SBV is seeking to merge five of the most financially sound banks in the country. SBV officials acknowledge that dealing with banks' bad debts also depend on what is happening in the real sector. The real estate sector remains a problem as property prices continued to soften in 1Q13 and the market remains frozen. An estimated 10%–12% of NPLs come from real estate lending.

With regard to crisis management protocols, the MOF, SBV, and State Securities Commission (SSC) have signed MOUs pledging cooperation with one another. Furthermore, the country has instituted safety nets, deposit insurance, and intensive supervision to counter any potential crisis.

The SSC, established in 1996, oversees and regulates securities trading on the country's two official exchanges: the Ho Chi Minh Stock Exchange and the Hanoi Stock Exchange. The SSC officially remains part of the MOF.

For capacity building, the SBV has its own training department and research institute, which also trains banking executives.

4.2 Capital Adequacy and Liquidity Management

The GFC exposed a number of weaknesses in the global regulatory framework and in banks' own internal risk management systems. To prevent a recurrence of the negative impacts on the global economy, the Basel Committee on Banking Supervision agreed in 2010–2011 on the Third Basel Accord (Basel III), with the latest installment of the so-called Basel Accords developed to address the deficiencies in financial regulation revealed by crisis.

The key principles addressed by Basel III concern capital requirements, leverage ratios, and liquidity requirements. Specifically, Basel III requires banks to hold 4.5% of common equity (up from 2% under Basel II) and 6% of Tier 1 capital (up from 4% under Basel II) as a ratio of risk-weighted assets (RWAs). Tier 1 capital refers to a bank's core capital, consisting primarily of common stock and retained earnings, whereas the risk weights contain credit, market, and operational risk measures.

In addition to the minimum capital requirements, Basel III introduced a mandatory capital conservation buffer of 2.5% and a discretionary countercyclical buffer of between zero and 2.5% of common equity.

Since excessive leverage was seen as a major cause of the crisis and its consequent severity, Basel III introduced a minimum leverage ratio to prevent the build-up of too much leverage. This leverage ratio is calculated by dividing Tier 1 capital by the bank's (unweighted) average total consolidated assets, with banks required to maintain the leverage ratio at not less than 3%.

To address liquidity risk, which emanates from maturity mismatches as banks fund illiquid long-term assets with very short-term debt, Basel III introduced minimum liquidity requirements. These have two components: (i) the liquidity coverage ratio, in which banks need to hold sufficient high-quality liquid assets to cover total net cash outflows over 30 days; and (ii) the net stable funding ratio, in which banks' available amount of stable funding must exceed the required amount of stable funding over a 1-year period of extended stress.

Basel III rules were originally scheduled to be introduced between 2013 and 2015. Changes in the rules made effective on 7 January 2013 included extending the implementation period until 2019.

The complexity of the new Basel III regulations—in addition to disputes regarding liquidity requirements, added cost burden on banks, expected toll on economic growth, and lack of an Asian perspective—have put the reform measures under a cloud of controversy. The questions of Basel III's relevance to Asian conditions, practices, and rising funding needs for its regional rebalancing efforts have become especially acute, leading to widening calls for an "Asian voice" in global banking regulatory reforms.

Asia largely sees Basel III requirements as primarily designed to address the problems of the developed economies of the US and Europe, with Asian voices underrepresented in the decision-making forums leading to their adoption. The new rules aim to resolve the undercapitalization and over-leverage of the wholesale banking model prevalent in the advanced economies. This contrasts with the prevailing retail banking model in Asia, in which heavy reliance on equity capitalization allows Asian banks to readily meet Basel III capital requirements. Their continued compliance is not so clear going forward, however, as Asia's continuous growth would put increasing demand on the credit supplied by a bank-dominated financial system.

Furthermore, banking systems across Asia are at different stages of development and Asian countries have diverse national development objectives. As part of their national goals, for instance, Asian economies have responded to the post-crisis economic slowdown in their advanced trading partners by shifting toward domestic- and regionally-driven growth, which requires funding. The region's developing capital markets also make it difficult and costly for Asian banks to meet the liquidity standards of Basel III due to an insufficient or non-existent supply of local government bonds.

Another issue concerns the risk weights used to compute minimum capital. Asian banks are disadvantaged against their Western counterparts since the risk weights rely on sovereign credit ratings and most Asian countries have lower ratings. In addition, Asian banks use a lesssophisticated standard model for risk weighting while their Western counterparts rely on internal risk-based models. As a result, Asian banks will have to set aside more capital to support the same amount of risky assets, such as loans and bonds.

In summary, the calls for an Asian voice in global financial reforms have been rising amid the perception that Basel III, among other initiatives, has become a "one-size-fits-all" solution that may not be suitable for many Asian economies, with their varied financial systems, stages of development, and banking industry practices. On the other hand, the strengths of Basel III include its notable attempt to standardize minimum capital and liquidity norms for global banking. Its main weakness is a heavy reliance on prescriptive measures that do not allow for flexibility and prudential discretion on the part of emerging economy banks and regulators in support of specific national development agendas.

Indicators	Basel III	Basel II
Minimum Tier 1 capital requirement	6% of risk-weighted assets (RWA)	4% RWA
Minimum common equity requirement	4.5%–7.0% RWA depending on countercyclical buffer	2% RWA
Countercyclical buffer Minimum capital adequacy ratio	Zero to 2.5% RWA 10.5% (including the capital conservation buffer)	None 8% RWA
Leverage ratio	Minimum ratio of common equity to exposures of 3%. This ratio will be tested before a mandatory leverage ratio is introduced in January 2018.	None
Liquidity ratio	There are two requirements – a liquidity coverage ratio (LCR, to be determined in 2015), and a net stable funding ratio (NSFR, to be determined in 2018).	None

Table 2: Selected Policy Changes in Basel II and III

Source: Liu and Moshirian (Forthcoming).

The table below summarizes the status of banking reform implementation in the five BCLMV countries. While most BCLMV countries have adopted Basel I regulations, none have fully adopted Basel II rules and none have started implementation of Basel III reforms.

	Basel Accords		
	Basel I	Basel II	Basel III
Brunei Darussalam	Adopted	Deferred	Not started
Cambodia	Adopted	Modified solvency ratio	Not started
Lao People's Democratic Republic	Implementing	Not started	Not started
Myanmar	Not started	Not started	Not started
Viet Nam	Adopted	Under consideration	Not started

Table 3: Banking Reforms in BCLMV Countries

Source: Authors' compilation.

Brunei Darussalam. Basel I core principles have been adopted but Basel II implementation has been deferred. All banks are Basel III compliant as far as the Tier 1 capital requirement is concerned since the capital funds of the banks in Brunei Darussalam consist solely of common equity. AMBD also closely observes the standards of the International Organization of Securities Commissions (IOSCO) in implementing its securities regulatory functions.

The capital adequacy of banks in September 2011 averaged 19.4% based on regulatory capital to RWA, and 20% based on Tier 1 capital to RWA.

Cambodia. The country's banking system dominates the economy with 32 commercial banks (including 10 foreign branches and 19 subsidiaries) empowered with universal banking functions but mainly doing traditional banking such as deposits, loans, and payment systems. Prudential regulations on liquidity and minimum capital have been tightened, with capital requirements raised from \$13 million to over \$35 million.

The implementation of Basel II rules are in a state of transition, whereby the NBC is using a "modified" 15% solvency ratio. It is also transitioning to a risk-based and forward-looking supervision model.

Banks are starting to comply with Basel II rules. The country has 32 commercial banks, including seven specialized banks that can only lend and do not have deposit-taking functions. Four or five of the largest banks account for about 70% of total loans and deposits. The government allows 100% foreign ownership of banks.

Banks have their own LTV policies and maintain a risk-management-based loan portfolio. Stress-testing is also conducted. Aside from mortgage lending, bank lending goes to the other leading sectors of the economy: tourism, agriculture (mainly rice), garments, and other manufacturing.

There is no insurance on deposits. The government regulates banks through loans, specifically by limiting access to financing and through collateral, which is based on the value of the asset

being financed (the assets should be 50% of the value of the collateral). Real estate values, the main form of collateral, declined in 2008–10 before recovering in 2011.

SMEs have no direct access to bank borrowings and instead resort to microfinance where interest rates can reach as high as 30% per annum.

The Lao People's Democratic Republic. The country's banking system is still implementing Basel I reform measures and is in the process of adopting Basel II reforms. The lagging pace of implementation is ascribed to the state of the supervision system itself, which needs to be developed since commercial banks are also unprepared in terms of expertise of staff and technical systems.

Myanmar. The banking system has not yet started implementing Basel I reforms and the capital adequacy ratio (CAR) in use does not meet international standards. The loan classification system in use also does not meet international standards and the rating system for banks is very basic.

Viet Nam. The central bank has already adopted Basel I reform and is considering Basel II reforms. The SBV intends to "do it step by step, depending on banks' and the market conditions." It is considering Basel III principles, electing to "pick up some of these principles, such as liquidity ratio." It has designed a roadmap for implementing Basel II reforms and is now in discussions with banks, which consider the proposed requirements to be costly. An informal group of 20 persons is in charge of the roadmap, whose development was funded through technical assistance from the Japan International Cooperation Agency (JICA).

4.3 Macroprudential Surveillance

One core function of central banks is to maintain the stability and soundness of the financial system. This macroprudential responsibility stands in contrast to their role as bank supervisors, which is more microprudential in scope. It is understood that financial stability requires stable banks and a sound banking system. The experience of recent financial crises, however, has provided the lesson that the soundness of individual banks is not enough to maintain the financial stability of the entire system.

According to the BIS: "[The GFC] has prompted a careful review of a wide range of policy areas. In many cases, microprudential supervision failed to ensure that financial institutions had sufficient capital and liquidity to cope with the shock. The efficacy of monetary policy in responding to system-wide financial risk in an environment of stable inflation was, and still is, under debate. The issue of how to define and develop the macroprudential element of financial stability policy has attracted particular attention. Policymakers broadly agree that the purpose of macroprudential policy is to reduce systemic risk, strengthening the financial system against shocks and helping it to continue functioning stably without emergency support on the scale that was extended in the crisis" (BIS 2010). The BIS further defines systemic risk as "a risk of disruption to financial services that is caused by an impairment of all or parts of the financial system and has the potential to have serious negative consequences for the real economy."

Davis (1999) defines macroprudential surveillance as the "monitoring of conjunctural and structural trends in financial markets so as to give warning of the approach of financial instability..." Meanwhile, the FSB, IMF, and BIS (2011) define macroprudential policy as "a policy that uses primarily prudential tools to limit systemic or system-wide financial risk, thereby limiting the incidence of disruptions in the provision of key financial services that can have serious consequences for the real economy..."

Practitioners utilize "the theory of financial instability and the experience of financial crises in the past... to enable meaningful use to be made of financial and macroeconomic data in macroprudential surveillance," according to Davis (1999). The research data include econometric forecasts and current information. Macroprudential surveillance is not mechanistic, as "detailed knowledge of the sequence of events in past crises, both directly and as encapsulated in theory, is a sine qua non to interpreting the data. In addition, there is a need for development of broad information on what constitutes normal conditions in an economy, as well as the patterns that have often preceded financial crises in the past both domestically and internationally."

Adding impetus to the progress in macroprudential surveillance, the IMF (2001) helped develop and distribute so-called macroprudential indicators (MPIs), which are "indicators of the health and stability of financial systems." These MPIs "will be critical in producing reliable assessments of the strengths and vulnerabilities of financial systems as part of IMF surveillance, and to enhancing disclosure of key financial information to markets." The lack of consensus on a core set of MPIs has been a major difficulty in this endeavor. Furthermore, "[the] ability to monitor financial soundness presupposes the existence of indicators that can be used as a basis for analyzing the current health and stability of the financial system."

These indicators comprise both aggregated microprudential indicators of the health of individual financial institutions and macroeconomic variables associated with financial system soundness. A set of indicators that the IMF has identified through its financial sector surveillance, technical assistance, and program work over the years is enumerated in the table below.

Aggregated Microprudential Indicators	Macroeconomic Indicators	
Capital adequacy	Economic growth	
Aggregate capital ratios	Aggregate growth rates	
Frequency distribution of capital ratios	Sectoral slumps	
Asset quality	Balance of payments	
Lending institution	Current account deficit	
Sectoral credit concentration	Foreign exchange reserve adequacy	
Foreign currency-denominated lending	External debt (including maturity structure)	

Table 4: Summary of Macroprudential Indicators

Table 4: Continued

Aggregated Microprudential Indicators	Macroeconomic Indicators
NPLs and provisions Loans to loss-making public sector entities	Terms of trade Composition and maturity of capital
	flows
Risk profile of assets	Inflation
Connected lending	Volatility in inflation
Leverage ratios Borrowing entity	Interest and exchange rates Volatility in interest and exchange rates
Debt-to-equity ratios	Level of domestic real interest rates
Corporate profitability	Exchange rate sustainability
Other indicators of corporate conditions	Exchange rate guarantees
Household indebtedness	Lending and asset price booms
Management soundness Expense ratios Earnings per employee Growth in the number of financial institutions	Lending booms Asset price booms Contagion effects Trade spillovers
Earnings and profitability	Financial market correlation
Return on assets	Other factors
Return on equity	Directed lending and investment Government recourse to the
Income and expense ratios Structural profitability indicators	banking system Arrears in the economy
Liquidity Central bank credit to financial institutions Segmentation of interbank rates Deposits in relation to monetary aggregates Loans-to-deposits ratios Maturity structure of assets and liabilities (liquid asset ratios) Measures of secondary market liquidity	
Sensitivity to market risk Foreign exchange risk Interest rate risk Equity price risk Commodity price risk	
Market-based indicators Market prices of financial instruments, including equity Indicators of excess yields Credit ratings Sovereign yield spreads	

Source: IMF (2001).

In an alternative approach, Hahm, Shin, and Shin (2012) proposed using the growth in banks' noncore liabilities as a predictor of crises. Banks, as the main financial intermediaries in developing economies, source their funding from retail deposits of households. However, retail deposits grow in line with the economy's expansion, so when credit is growing faster than deposits, banks turn to other sources of funding. By classifying retail deposits as banks' core liabilities and the remaining components of bank funding as noncore liabilities, the ratio of noncore to core liabilities reflects the pace of credit growth relative to trend. A rising ratio would indicate the economy's rising risk premia. The authors further observed that foreign liabilities of banks in emerging economies comprise a major component of their noncore liabilities, as the underdeveloped domestic wholesale funding market cannot support the rapid growth in bank lending.

According to Hahm, Shin, and Shin (2012), measures of the noncore bank liability ratio "have significant predictive power for currency crises and credit crises," based on the information contained in the banking sector's liabilities to the foreign sector. In addition, "the noncore bank liability ratio has independent predictive power over the much better-known and debated credit-to-GDP ratio."

They contend, that their "findings suggest that, at least in emerging and developing economies, noncore bank liabilities may be usefully monitored as a complementary measure to the credit to GDP ratio in gauging the stage of financial cycles and the build-up of financial risk."

Meanwhile, most developing economies, including the BCLMV countries, suffer from a dearth of available data that precludes effective macroprudential surveillance as well as the proposed monitoring of noncore bank liabilities.

In response, ADB's Office of Regional Economic Integration (OREI) has embarked on a project to build a database on banks' noncore liabilities in the BCLMV countries, while a similar database is also being built for Indonesia, Malaysia, the Philippines, Thailand, and Singapore. In this database, core liabilities refer to liabilities of the financial intermediary sector to non-bank domestic creditors, while noncore liabilities include foreign exchange liabilities of the financial intermediary sector, and claims held by intermediaries on other intermediaries.

As studies have shown, an alarming accumulation of noncore liabilities could signal mounting vulnerability to both liquidity and currency crises, and therefore, it is important to monitor the movements and size of the financial intermediary sector's noncore liabilities.

These core and noncore liabilities should be further classified by type of holder to determine whether the holders of the liabilities are the ultimate domestic creditors. Examples of core and noncore liabilities, classified further in terms of degree of liquidity, are given in the table below.

	Core Liability	Intermediate	Noncore Liability
Highly Liquid	Cash Demand deposits (households)	Demand deposits (non-financial corporates)	Repos Call Ioans Short-term FX bank debt
Intermediate	Time deposit and Certificates of Deposit (households)	Time deposit and Certificates of Deposit (non-financial corporates)	Time deposit and Certificates of Deposit (banks and securities firms)
Illiquid	Trust accounts (households) Covered bonds (households)	Trust accounts (non-financial corporates)	Long-term bank debt securities (banks and securities firms) ABSs and MBSs

Table 5: Core vs. Noncore Liabilities (by degree of liquidity)

ABSs = asset-backed securities, MBSs = mortgage-backed securities. *Source:* Shin and Shin (2011).

It is important that data are disaggregated by type of holder, such as households and nonfinancial corporates. For each of these holders, wherever appropriate, the relevant indicators include cash, demand deposits, time deposits, certificates of deposit, trust accounts, covered bonds, repurchase agreements, call loans, short-term foreign exchange bank debt, long-term bank debt securities, and asset-backed securities and mortgage-backed securities.

ADB researchers surveyed the central bank websites of the BCLMV countries to collect publicly available data on noncore liabilities. As summarized in the table below, most of the data being sought were not publicly available. Building and regularly updating a database of these indicators for the BCLMV countries would be a useful exercise as these countries gradually emerge as important players in global financial markets. Monitoring the noncore liabilities of their financial intermediary sectors can help detect emerging liquidity and currency risks before they turn into a full-blown financial crisis.

	Brunei Darussalam	Cambodia	Lao People's Democratic Republic	Myanmar	Viet Nam
Core Liabilities Cash	х		х	x	x
Demand deposits (households)	х	х	х	х	x
Time deposits (households)	х	х	х	х	x
CDs (households)	х	x	x	x	x
Trust accounts (households)	х	х	х	х	x
Covered bonds (households)	x	×	x	x	x
Intermediate					
Demand deposits (non-financial corporates)	x	x	х	х	x
Time deposit (non-financial corporates)	х	x	x	x	x
CDs (non-financial corporates)	х	х	х	х	x
Trust accounts (non-financial corporates)	x	х	x	x	x
Noncore Liabilities					
Repos	Х	Х	Х	х	х
Call Ioans	х	х	х	x	х
Short-term FX bank debt	x	x	x	x	x
Time deposits (banks and securities firms)	x	x	x	x	x
CDs (banks and securities firms)	x	x	x	x	x
Long-term bank debt Securities (banks and securities firms)	×	x	x	x	x
ABSs and MBSs	x	x	x	x	x

Table 6: Availability of Data on Core and Noncore Liabilities of the Financial Intermediary Sector in BCLMV Countries

ABSs = asset-backed securities, BCLMV = (Brunei Darussalam, Cambodia, the Lao People's Democratic Republic, Myanmar and Viet Nam), CDs = certificates of deposit, FX = foreign exchange, MBSs = mortgage-backed securities.

Capital Flows and Noncore Liabilities

Banks in emerging Asian economies, including those in BCLMV countries, intermediate not only internal funds available domestically but, just as importantly, they intermediate a large amount of cross-border capital flows. Due to their procyclical nature, these capital flows have important implications for emerging Asian economies' financial stability. Their high volatility, furthermore, adds a dimension of challenge to regulatory, political, and economic policymaking.

The volatile feature of these flows was underscored in the paper of Cho and Rhee (2013) that compared trends in capital flows in 10 Asian economies before and after the GFC.⁶ The trend analysis indicated a sharp rise in inflows to over \$1.4 trillion in 2007 and a sharp plunge to less than \$200 billion during the worst of the GFC in both 2008 and 2009. As the US Federal Reserve launched its large-scale asset purchases, otherwise known as quantitative easing, the capital flows into Asia sharply rebounded to pre-crisis levels in 2010 and 2011, before easing again in late 2011 in the wake of the worsening European sovereign debt crisis. In terms of the composition of capital inflows, bank loans exceeded their pre-crisis levels and became the main source of capital inflows after the GFC.

On the other hand, most of the BCLMV countries were spared these volatile capital flows due to their serendipitous low degree of financial integration with the region and the rest of the world, lack of capital account openness, and relatively undeveloped capital markets and financial sectors. Relative exceptions would be Cambodia, whose dollarized economy suffered credit outflows in 2009, and Viet Nam, which alone among the five BCLMV countries is included in the "frontier markets" indices of major market data aggregators.

While these five countries under study have yet to gain the capacity and develop the institutions required to absorb and benefit from the inflow of capital from abroad, it would be instructional to thresh out the linkages between noncore liabilities and capital flow management.

The banking sector, in particular, traditionally intermediates the wholesale funding variety of capital flows, which are highly procyclical and tend to quickly reverse when adverse conditions emerge. These debt-type inflows can thus become significant transmitters of instability.

The impacts of bank-related capital flows are determined by a number of factors, including how domestic banks manage their balance sheets and source out funds, and how multinational banks organize themselves.⁷

⁶ The 10 Asian economies are the PRC; Hong Kong, China; India; Indonesia; Japan; the Republic of Korea; the Philippines; Singapore; Taipei, China; and Thailand.

⁷ A fuller discussion of these issues may be seen in Committee on International Economic Policy and Reform. 2012. Banks and Cross-Border Capital Flows: Policy Challenges and Regulatory Responses. Washington, DC: Brookings Institution.

Bank lending is highly procyclical and this procyclicality is further amplified by the slack in banks' balance sheets, which refers to banks' ability to take on additional risks. When the pool of available retail deposits becomes exhausted during a credit boom, banks will turn to alternative sources such as wholesale lending banks, whose support typically comes in the form of cross-border funding. Short-term wholesale funding from foreign creditor banks, which become the noncore liabilities of domestic banks, are more volatile than retail funding and would typically fall in the boxes of Tables 6 and 7 that encompasses repos, call loans, and short-term foreign exchange bank debt, as well as time deposits and certificates of deposit (upper rightmost boxes in Table 6 and lowermost boxes in Table 7).

How large banks organize their international operations also has implications for capital flows and financial stability. In particular, international banks with a decision-making process that remains centered at the home country headquarters, and with funding pooled from each country source and redistributed from the headquarters, can transmit shocks more directly and quickly through changes in affiliates' funding. Foreign banks, operating either as affiliates or branches, can easily migrate to cross-border, noncore funding for their credit expansion needs. Another source of funding would be inter-office funding channeled by the parent bank through its subsidiaries.

Capital flows, in addition, can affect the balance sheet of custodian banks in a distinctive way. A custodian refers to a bank or financial institution that holds securities on behalf of investors. Custodian banks' basic tasks include safekeeping securities, accepting or rejecting traded securities, asset servicing (processing of rights, collecting principal, interest, or dividend payments), and cash management and foreign exchange services. Global custodians whose customers invest in foreign securities contract with custodians in foreign countries, known as "sub-custodians," to provide local custody services. Providing custodian services can affect banks' cash flows and liquidity.⁸ Gross inflows and outflows from customers' payment traffic can be very large and volatile. The impact of custodian services on a bank's balance sheet would be felt more on the core liabilities such as cash and demand deposits (upper leftmost box in Table 6 and uppermost boxes of Table 7.

Brunei Darussalam. A macroprudential surveillance system is still being set up. A financial stability committee exists as an ad hoc group. AMBD does not specify LTV ratios or DTI rules, nor does it mandate collateral values for mortgages, preferring to merely monitor these matters. Onsite and offsite monitoring is done on a regular basis; a complete onsite examination, lasting 3–6 months has just been completed. It uses the CAMEL framework to evaluate banks.

Cambodia. NBC is just starting to implement macroprudential surveillance. No specific department or task force has been designated for this purpose. For disclosure and transparency purposes, banks are required to report and publish in June of every year their audited financial statements on their websites and in newspapers of general circulation. NBC is developing an offsite inspection manual that is scheduled to be implemented in the middle of

⁸ These issues are discussed in BIS (2008).

2013. NBC branches conduct onsite inspection in all provinces on a sample basis, using the CAMEL methodology in its bank examinations.

As part of macroprudential surveillance, the Supreme National Economic Council and the MEF look at the monetary survey. Over 95% of deposits in the country, consisting of savings and demand and time deposits, are in US dollars, which creates a problem with intermediation and the resulting non-securitized intervention by economic managers.

With an open economy and a liberal exchange rate system, the US dollar has effectively become a second national currency existing alongside the national currency, the riel, which was introduced in 1955. The dollar's dominant use over the riel, however, has presented a number of challenges to Cambodia's monetary authorities. Capannelli and Menon (2010) pointed out that as a result of dollarization, Cambodia's monetary authorities cannot directly influence the money supply and cannot conduct exchange rate policy effectively. Consequently, the burden for macroeconomic adjustments in Cambodia falls mainly on fiscal policy.

Signs of financial deepening, based on the ratio of the M2 money supply to GDP, are evident. In particular, the M2 money supply has been growing on the back of increasing foreign currency deposits and credit growth, and with the entry of new banks. However, banks have been limited to short-term lending. Rates on deposits are at 6% per annum, while lending rates range from 10% to 15%.

Lao People's Democratic Republic. A BOL unit has been set up to work on the planned macroprudential surveillance system, which is expected to be set up by 2015. The BOL is learning from the experiences of the Bank of Thailand and Bank Indonesia in this endeavor.

Myanmar. The CBM's priority plans for capacity building include macroprudential and crisis prevention and resolution, risk management systems (since credit risk is no longer the sole risk in an evolving financial system), and risk-based supervision. In general, the plan is to follow the ASEAN framework plan for economic integration by 2015.

Viet Nam. The SBV already has an early warning system and uses macroprudential indicators. A financial sector assessment program being conducted by the IMF is expected to be launched by the end of 2013, with a specific unit in the SBV in charge of this effort. (There is no specific SBV department in charge of macro surveillance, however.) The SBV also has a credit bureau, the Credit Information Center, which was established in 2008.

4.4 Disclosure and Transparency

Disclosure and transparency form one of the six major principles of corporate governance of the OECD and one of the BIS' six principles for enhancing corporate governance.

"The corporate governance framework should ensure that timely and accurate disclosure is made on all material matters regarding the corporation, including the financial situation, performance, ownership, and governance of the company," according to the OECD (2004).

"A strong disclosure regime that promotes real transparency is a pivotal feature of marketbased monitoring of companies and is central to shareholders' ability to exercise their ownership rights on an informed basis," the OECD (2004) further states. "A strong disclosure regime can help to attract capital and maintain confidence in the capital markets. By contrast, weak disclosure and non-transparent practices can contribute to unethical behaviour and to a loss of market integrity at great cost, not just to the company and its shareholders but also to the economy as a whole." OECD (2004)

For its part, the BIS (2010) notes "Transparency is one tool to help emphasise and implement the main principles for good corporate governance." Principle 14 for enhancing corporate governance states: "The governance of the bank should be adequately transparent to its shareholders, depositors, other relevant stakeholders, and market participants... The objective of transparency in the area of corporate governance is therefore to provide these parties, consistent with national law and supervisory practice, with key information necessary to enable them to assess the effectiveness of the board and senior management in governing the bank... Such disclosure should be proportionate to the size, complexity, structure, economic significance, and risk profile of the bank."

Empirical studies tend to support the importance of disclosure and transparency to investors and, consequently, to the flow of capital. The IMF (2001) commented: "A lack of transparency was a feature of the build-up to the Mexican crisis of 1994–95 and of the emerging market crises of 1997–98. In these crises, markets were kept in the dark about important developments and became first uncertain and then unnerved as a host of interrelated problems became known. Inadequate economic data, hidden weaknesses in financial systems, and a lack of clarity about government policies and policy formulation contributed to a loss of confidence that ultimately threatened to undermine global stability... Transparency and candor are particularly important in today's environment of substantially increased private capital movements and countries' growing integration with international capital markets."

In a study, Gelos and Wei (2002) investigated the effect of transparency in developing countries on the level of investment by international institutional investors. They enumerated several noteworthy findings. "First, we find relatively clear evidence that international funds prefer to hold more assets in more transparent markets. Second, on the other hand, herding among funds is more prevalent in less transparent countries. Third, there is some modest evidence that during a crisis, international investors tend to flee more opaque markets." Here, "herding" is defined as "funds taking investment decisions which they would not take if they did not observe other funds taking them." Their findings suggest that improved transparency can reduce the sudden reversal of capital flows during a crisis and thus enhance the stability of domestic financial markets in a developing country.

Box 1: Organisation for Economic Co-operation and Development Principle on Disclosure and Transparency

The corporate governance framework should ensure that timely and accurate disclosure is made on all material matters regarding the corporation, including the financial situation, performance, ownership, and governance of the company.

A. Disclosure should include, but not be limited to, material information on:

1. The financial and operating results of the company

- 2. Company objectives
- 3. Major share ownership and voting rights
- 4. Remuneration policy for members of the board and key executives, and information about board members, including their qualifications, the selection process, other company directorships, and whether they are regarded as independent by the board
- 5. Related party transactions
- 6. Foreseeable risk factors
- 7. Issues regarding employees and other stakeholders
- 8. Governance structures and policies, in particular, the content of any corporate governance code or policy and the process by which it is implemented.

B. Information should be prepared and disclosed in accordance with high-quality standards of accounting and financial and non-financial disclosure.

C. An annual audit should be conducted by an independent, competent, and qualified auditor in order to provide external and objective assurances to the board and shareholders that the financial statements fairly represent the financial position and performance of the company in all material respects.

D. External auditors should be accountable to the shareholders and owe a duty to the company to exercise due professional care in the conduct of the audit.

E. Channels for disseminating information should provide for equal, timely, and cost-efficient access to relevant information by users.

F. The corporate governance framework should be complemented by an effective approach that addresses and promotes the provision of analysis or advice by analysts, brokers, rating agencies and others that is relevant to decisions by investors and free from material conflicts of interest that might compromise the integrity of their analysis or advice.

Brunei Darussalam. Aware of the merits of public listing, the government is in the process of establishing a stock exchange under its capital markets development plan. Senior AMBD management is in charge of this plan. No timetable has been set.

AMBD has issued up to \$100 million of *sukuk* (Islamic bonds), with maturities of up to 1 year, mainly to set benchmarks and contribute to capital market development.

Cambodia. The development of the securities sector had been laid out in the *Financial Sector Development Strategy 2006–15* and the *Rectangular Strategy, Stage II*. Under these plans, the government considers the securities sector as an essential dimension of the financial system, providing the mechanism for mobilizing all sorts of financial resources for investment projects and to diversify the sources of economic growth. Moreover, the existence of the securities market will promote accountability, transparency, good corporate governance, and efficiency in the operations of firms, especially those engaging in public offers.

These objectives have been carried over into the newer *Financial Development Strategy 2011–20*. Capital market development is seen as a catalyst in promoting good corporate governance in the public sector, as companies are required to have 3 years of audited financial statements prior to listing. Market development will be conducted in three overlapping stages, with the first stage (2011–14) focused on bond market development, the middle stage (2014–17) focused on equities, and the last stage (2017–20) focused on the introduction of derivatives.

The SECC, which supervises the securities sector in the country, has six operating departments: Securities Market Supervision; Securities Intermediaries Supervision; Legal Affairs; Securities Issuance Supervision; Research, Training, Securities, Market Development and International Relations; and Administration and Finance. It has the power to arrest without the prosecutor's opinion, and to hold the suspect for 48 hours. It has recorded one case in 2010 for an illegal security offering. It uses criminal laws to prosecute. The sanctions it imposes fall under two major categories: administrative and criminal.

The SECC further supervises 13 securities companies, consisting of seven underwriters (five of which are foreign), two dealers (one foreign), and four brokers (zero foreign).

It has a 30-investor rule as the threshold: companies with more than 30 investors are required to engage in a public offering of shares.

Under the law, listed public companies are differentiated from listed private companies in terms of corporate governance rules. Public companies cannot have more than seven directors, at least one of which must be independent, while private companies can have between seven and 15 directors, of which 20% must be independent.

The SECC foresees the local market being integrated with the ASEAN Exchanges, an existing collaboration among four national exchanges (Malaysia, the Philippines, Singapore, and Thailand). The SECC's top priorities are listing promotion; IPO evaluation, processing, and procedures; corporate governance; market surveillance systems; and market investigation. A team has been set up to spearhead these efforts.

The Lao People's Democratic Republic. Disclosure requirements consist of the publication of annual audited financial statements within the first quarter each year, in addition to the submission of monthly statements.

A new securities bill passed in December 2012 waits to be signed into law. Under the law, the LSX, with status as a self-regulatory organization, will be in charge of monitoring and regulating

corporate disclosure, with the SEC in charge of enforcing irregular disclosures. The LSX currently only trades the stocks of two companies: a bank (BCEL) and a power utility (EDL). An SEC committee has drafted a long-term development plan for the domestic capital market to be released and implemented by end of 2013. The Lao PDR's accession to the WTO in 2012 is seen as a plus for the country's capital market development.

The government permits 100% foreign-owned banks to operate in the country and sets a 20% foreign ownership limit for industrial nonbank companies. The 5% threshold rule is enforced.

The MOF is in charge of the government bond market. Auctions are conducted once a year, although the government desires to eventually conduct one auction every month. Investors hold these treasuries to maturity as a secondary market for treasuries is still under study. The legal framework for this secondary market is lacking and policymakers are still looking for a model to follow and are open to receiving technical assistance on this matter. Clearing is done by the BOL.

The government issues Treasury securities: Treasury bills are issued to cover the fiscal deficit while bonds with maturities of up to 5 years have also been issued. There are no corporate bonds. Under discussion is whether the LSX will handle the secondary market for bonds. The MOF previously considered the issuance of cross-currency asset-backed securities to fund its hydro project. However, this funding alternative did not progress because of issues regarding the transfer of royalties as collateral for the securities.

Tax incentives are provided to develop the capital market, mainly consisting of a 5-year exemption from the 24% capital gains rate. Additional fiscal incentives are not possible at present due to the government's primary goal of reducing poverty, which entails financing social programs, subsidies for SOEs, and bank lending to certain strategic sectors.

Myanmar. The country has no functioning capital market, a situation that is to be addressed by the proposed Security Exchange Law, which is now awaiting passage in Parliament. This law will also set up a Securities and Exchange Commission.

This process has been years in the making. A steering committee was created in 1996 to set up a stock exchange, with the CBM taking the lead. In June of that year, the Myanmar Securities Exchange Center Co., Ltd (MSEC) was established as a 50:50 joint venture between the state-owned Myanmar Economic Bank (MEB), the country's largest bank, and Japan's Daiwa Institute of Research Ltd. MSEC's business activities include brokering, dealing, underwriting, and distributing securities, as well as research and consultancy services. It also sells Myanmar Treasury bonds as an agent for the CBM.

In May 2012, the CBM, Daiwa Research Institute, and Tokyo Stock Exchange signed a technical aid agreement to develop the country's stock exchange by 2015. Daiwa will help train the stock exchange's workforce and advise on the set-up of the necessary IT systems, while the TSE will help establish the stock exchange's rules and operating standards.

32 | Working Paper Series on Regional Economic Integration No. 134

Viet Nam. Public disclosure requirements compel companies to report to the SSC and Hanoi Stock Exchange (HSX), and also to provide their disclosure reports on their websites. As the SSC issues new guidelines on filing rules, it is implementing a system for the electronic transmission of disclosure reports. However, as some companies have limited access to the internet, almost all listed companies still either send hard copies or scanned copies. Authentic verification remains an issue for electronic filing. The new system, the Information Disclosure System (IDS), was set-up with the help of the Luxembourg Agency for Development Cooperation and the FPT Information System. It is slated to be operational by the end of 2013. The initial stage of implementation will be for public companies. FPT Information System is a local IT solutions provider that is listed on the Ho Chi Minh Stock Exchange.

The fine for late or inaccurate disclosure is D20 million, an offense that occurs very often, with around 15 companies investigated every year. Circular 52 classifies disclosures into three main groupings: regular (about 10 items, including audited financial statements, annual reports, corporate governance certifications, and BOD resolutions); unusual; and as required by the regulator. These groupings adhere to international standards on disclosure.

Decree 58 covers foreign enterprises' issuance of shares, mandating that foreign companies can list only those shares issued in Viet Nam. To date, no foreign enterprise has availed of this listing rule.

Foreign ownership is limited to 49% for listed non-banking companies and 30% for banks. A 5% ownership threshold is mandated for foreign and local investors, whereby the investor or group must file a disclosure report. Each 5% increment thereafter, up to 25%, requires a corresponding disclosure. Ownership above 25% requires approval from the SSC.

A two-tier market surveillance system is maintained; the primary tier covers surveillance at the stock exchange level and the second tier at the SSC level. The parameters for market surveillance have been developed by the stock exchange. For its part, the SSC administers compliance requirements over the stock exchange and monitors transactions conducted on the exchange. Illegal trades fall under criminal proceedings and carry penalties plus a prison sentence. Administrative cases currently carry a fine of up to D2 billion.⁹

In its investigations, the SSC can access an investor's ID from security brokers and documents from the settlement bank, but it has no direct access to investor communications (e.g., e-mails, phone records). The investigation department then refers the case to the inspection department, which forms an examination group, which if needed then refers the case either to prosecutors for a criminal case or the relevant SSC department for an administrative case. It takes a maximum 10 days for a case to be elevated from the stock exchange to the SSC, where the surveillance department may process the case over a few months or even years for complex cases. For its part, the Ho Chi Minh Stock Exchange interviews, investigates, and asks listed companies for an explanation within 24 hours.

⁹ Decree No 108/2013/ND-CP, which took effect on 15 November 2013, raised maximum fines up to D2 billion (\$95,238) for institutions and D1 billion (\$47,619) for individuals—quadruple the prior rates.

Circular 183, which covers the establishment of mutual funds, was instituted in 2012. In the early part of 2013, the local financial industry had 23 closed-end funds and 2 open-end funds.

4.5 Capital Flow Management

Capital flows can be a double-edged sword for developing economies. On one hand, capital inflows into a developing economy can be a great boon to growth, financing developmental requirements, smoothing consumption patterns, diversifying risks, and expanding economic opportunities. On the other hand, capital flows can also create a host of problems.

"Although few deny the benefits of capital inflows for recipient economies, they often create problems especially if they are substantial and volatile," stated Kawai and Lamberte (2010). "Massive capital inflows," they continued, "can create too much bank lending, excessive investment, and speculative activities, which can lead to goods price inflation, asset market bubbles, and potential vulnerabilities in bank, household, and corporate balance sheets. Moreover, sudden stops or reversals in capital inflows could lead to a currency crisis, the bursting of asset price bubbles, investment collapse, banking sector stress, and economic difficulties."

Since recovering from the 1997–1998 Asian financial crisis, the region's developing economies have seen the return of capital inflows and a string of current account surpluses over a number of years. The more recent GFC that started with the collapse of the US subprime mortgage housing market left most Asian economies relatively unscathed, although their foreign trade inflows suffered due to the downturn in demand from developed economies.

These Asian economies have proved to be resilient, expanding in the post-crisis years while developed economies continue to either recover at a snail's pace or remain in a slump.

"There are signs that capital of a largely short-term nature is returning to Asia in a significant way, raising serious concern among policymakers in the region who are trying to prevent rapid appreciation of their currencies against the US dollar and to contain inflation and increases in asset prices to stabilize their economies and sustain the recovery," stated Kawai and Lamberte (2010). They added: "These recent developments suggest that managing capital inflows remains an important policy issue for many emerging market economies that needs to be studied rigorously and debated openly. Policymakers in the region are faced with questions on best policy responses and regional cooperation initiatives to utilize capital inflows while maintaining macroeconomic stability."

Brunei Darussalam. The combination of a healthy level of international reserves—a major necessity for a currency board system—as well as a perennially positive trade balance, a huge current account surplus that exceeds 40% of GDP, fiscal surpluses, no foreign debt, and the presence of flexible markets for goods and labor have rendered moot the management of capital flows in Brunei Darussalam.

Cambodia. As a dollarized economy, there are no restrictions on international capital flows, but local banks are not allowed to invest abroad. NBC admits this is not a good system of

34 | Working Paper Series on Regional Economic Integration No. 134

capital flow surveillance from the point of view of financial stability. This also makes monetary policy inutile and achieving price stability difficult. The government does not issue either Treasury bills or bonds, as it remains cheaper to borrow abroad at concessional rates. Debt management capacity thus has to be developed. The government also hopes to develop an early warning system under its 2020 financial sector development plan.

The Lao People's Democratic Republic. Total international reserves have surged in recent years, surpassing \$700 million in 2010, more than double the amount 5 years earlier. This has been fueled by the financial account balance—mainly consisting of direct investments and other investments from abroad. Experts note that investors from Viet Nam have been shifting funds to Lao PDR banks in anticipation of the depreciation of the Vietnamese dong. Capital flow management in the Lao PDR remains rudimentary.

Myanmar. The country is still in the process of attracting foreign investment and does not yet have a capital flow management system. Its funding has come mainly from concessional loans. However, attracted by Myanmar's ongoing reform and liberalization policies, as well as the lowest labor wages in the region, foreign firms have been rushing to search for investment opportunities.

Viet Nam. Although Viet Nam restricts capital inflows and outflows, the country enjoys robust inflows of capital, consisting mainly of FDI. The Ministry of Planning and Investment has the authority to grant foreign investment licenses for projects over \$1 million, while provincial authorities approve projects under \$1 million. Foreign individuals and organizations are allowed to purchase local securities, but any issuance or sale of securities by non-residents requires SBV approval. Furthermore, enterprises operating in the country are subject to annual overall ceilings on foreign borrowings.

From 1988 to 2011, registered FDI summed to almost \$230 billion. These capital flows once largely funded the country's current account deficit, but since 2005 remittances and portfolio inflows have played an increasing role in financing the trade deficit. With overseas remittances estimated at \$9 billion in 2012, the country ranks ninth among the world's largest remittance-receiving countries and territories. The current account surplus surged to over \$9 billion in 2012, in part due to low imports. With this, gross international reserves rose at end-February 2013 to more than 2.5 months of prospective imports of goods and nonfactor services. Based on official data, Viet Nam's international reserves stood at \$13.5 billion at the end of 2011, up from a low of \$12.5 billion in 2010, but still down from the recent high of almost \$24 billion in 2008.

5. Summary Findings and Recommendations

In sharp contrast to the 1997–1998 Asian financial crisis, the contagion effects of the recent GFC did not spill over into the financial sectors of ASEAN countries and left their economies relatively unharmed as well. The pain inflicted by the earlier Asian crisis, in terms of lost economic momentum, currency collapse, social unrest, and bank insolvencies, provided the impetus for countries in the region to embark on a series of economic, banking, and capital

market reform initiatives. The 1997–1998 Asian financial crisis drove the region's developing economies to see the globalization handwriting on the wall, and they opted to benefit from this trend by instituting the necessary reforms and pushing for greater regional cooperation.

This sense of urgency is sadly lacking in the aftermath of the recent GFC, which engulfed the developed economies and left Asia's developing economies in a relatively robust and resilient position. This larger crisis, however, affected the weakened advanced economies' demand for Asian exports. Asian economies, led by the PRC, turned inward with development programs aimed at re-balancing their economies by bolstering domestic consumption. The search for alternative markets is also pushing the agenda for greater regional integration in the form of the AEC. This, however, may prove to be too weak of an incentive for the BCLMV countries to accelerate economic and financial reforms.

As has been stated, the BCLMV countries differ widely in terms of political systems, stages of economic development, and cultures, among other characteristics. The major implication of this heterogeneity is that there is no one-size-fits-all solution for each country's response to the banking reform measures emanating from developed economies. Each of the BCLMV countries has its own national banking development agenda and timeframe for achieving its goals. A broad description of the priorities for financial sector development in the BCLMV countries is provided in the table below.

Box 2: Financial Sector Development Priorities in the BCLMV Countries

Brunei Darussalam

Enhance national payments system Set up a trade information bureau Establish a deposit insurance scheme under the MOF

Cambodia

Capacity building efforts focusing on the following:

- prepare for ASEAN integration (2015 deadline for the AEC)
- develop capital account monitoring system
- develop effective monetary policy (open market operation and benchmark interest rates)
- enhance banking supervision and regulation (onsite and offsite)
- strengthen payments system
- institute a modern information system

Lao People's Democratic Republic

- Priorities under the 2010–15 plan:
- develop the securities market
- institute a capital market strategy plan
- prepare for ASEAN financial integration in line with sustainable Basel II reforms

Also, transfer ownership of SOEs to the public (based on a MOF-drafted government decree)

Myanmar

Reform financial system (rewrite banking and financial laws) Grant independence to the Central Bank of Myanmar (CBM) New laws on financial institutions, foreign exchange management, and securities exchange Reform accounting and disclosures

Viet Nam

The government has issued two resolutions to boost economic growth and quickly resolve the real estate and NPL issue

Set up Viet Nam Asset Management Company (VAMC) and merge "five strong banks"

AEC = ASEAN Economic Community, ASEAN = Association of Southeast Asian Nations, BCLMV = (Brunei Darussalam, Cambodia, the Lao People's Democratic Republic, Myanmar and Viet Nam), MOF = Ministry of Finance, NPL = non-performing loan, SOEs = state-owned enterprises.

Source: Authors' compilation.

The main objective of the Basel reforms to fortify global financial stability clearly cannot take precedence over the BCLMV countries' national economic development aspirations. Adoption of these reforms will entail costs in terms of constraining economic progress. With the exception of Brunei Darussalam, the CLMV countries are in their catch-up phase of economic development marked by rapid growth. As these countries continue to grow, their financing requirements for infrastructure development will be substantial and continuously increasing.

Global financial reforms, especially the Basel reforms, may constrain the capacity of banking sectors in the BCLMV countries to provide their portion of the required funding.

Ultimately, however, the relevant dictum is that the financial sector must serve the needs of the real economy. But for it to do so effectively, financial sector development must go handin-hand with the economy's overall development. Financial development, including the advancement of the banking system, is a powerful determinant of economic growth. Part of this development is the regulatory regime and certain reforms are essential to enhancing its structure, such as establishing an independent regulatory and supervisory system, and ensuring compliance with international standards for bank supervision.

Given the underlying dissimilarities among the five BCLMV countries, the sequencing of financial reform measures should likewise diverge among them. Being at different stages of financial development, each country will have to set its own pace for opening up and reforming its financial sector. Brunei Darussalam, for example, presents a special case due to its oil-fueled high per capita income and lagging financial sector. While it has issued a small amount of *sukuk*, it has yet to establish a market for publicly traded securities, including stocks and bonds. In contrast, Cambodia, the Lao PDR, and Viet Nam have made strides in launching their own national stock exchanges.

The sequencing of which laws and regulations should be revised or enacted, and which institutional capacities need to be strengthened, is best decided by each country's set of policymakers. At the same time, this does not mean there is no need for the synchronization of implementation timeframes. It is crucial for the other five member countries of ASEAN, as well as for multilateral funding agencies, to push the BCLMV countries to accelerate their financial sector development programs. It would be risky, for example, for the AEC to be in general compliance with global regulatory reforms while allowing peripheral zones of non-compliance. There is the real danger that the BCLMV countries' continued rapid growth and development might outpace their regulatory capabilities, thus risking financial instability and planting the seeds of financial contagion in the region. Prolonged non-compliance would also unlock opportunities for regulatory arbitrage, which could become another source of instability within the region.

	Brunei Darussalam	Kingdom of Cambodia	Lao People's Democratic Republic	Republic of the Union of Myanmar	Socialist Republic of Viet Nam
Financial regulatory & supervisory system (independenc/resiliency)	No securities regulator	SECC under MEF	SECO and LSX are BOL units	CBM under MFR; no securities regulator	SSC under MOF
Capital adequacy & liquidity management	Adopted Basel I, deferred Basel II	Adopted Basel I, modified solvency ratio of Basel II	Implementing Basel I	Not started on Basel I	Adopted Basel I, mulling Basel II
Corporate governance & risk management	No CG for banks; limited disclosure for registered firms	Banks required to have independent board member and special committees for audit and risk management	CG contained in the Commercial Bank Law (c. 2006–2007)	CG rules to be addressed by proposed Financial Institutions Law	CG contained in Law on Enterprises, Banking Law 2010, Circular 121, and Decision 12
Macroprudential surveillance	System being set up	NBC just starting on it	BOL unit working on it	Included in CBM priorities	Already in operation
Transparency & disclosure	Plans to put up stock exchange	Plans laid out in Financial Sector Development Strategy 20062015	Securities Law, passed in December 2012, waits to be signed	Security Exchange Law, now waiting for parliament for approval	Two-tier market surveillance system is maintained; rules under Circular 52 and Decree 58
Capital flows management	Oil sales receipts main source of funding	Dollarized economy, no restrictions on capital flows	Rudimentary system	No formal system	MPI and SBV regulate flows

Table 7: Summary: Financial Monitoring	of BCLMV Countries
--	--------------------

BOL = Bank of Lao PDR, BCLMV = (Brunei Darussalam, Cambodia, the Lao People's Democratic Republic, Myanmar and Viet Nam), CBM = Central Bank of Myanmar, CG = corporate governance, LSX = Lao Securities Exchange, MEF = Cambodian Ministry of Economy and Finance, MFR = Ministry of Finance and Revenue, MOF = Ministry of Finance, MPI = Ministry of Planning and Investment, NBC = , SBV = State Bank of Viet Nam, SECC = Securities and Exchange Commission of Cambodia, SECO = Securities and Exchange Commission Office, SSC = State Securities Commission.

Source: Authors' compilation.

References

- Asian Development Bank. 2012. *Myanmar in Transition: Opportunities and Challenges*. Manila: Asian Development Bank.
- ———. 2013. The Road to ASEAN Financial Integration: A combined study on assessing the financial landscape and formulating milestones for monetary and financial integration in ASEAN. Manila: Asian Development Bank.
- Association of Southeast Asian Nation. 2008. ASEAN Economic Community Blueprint. Jakarta: Association of Southeast Asian Nations.
- Autoriti Monetari Brunei Darussalam. 2013. *Regulation of Lending Rates and Deposit Rates*. News Release. 8 March.
- Bank of International Settlement. 2008. Principles for Sound Liquidity Risk Management and Supervision. Basel: Bank for International Settlements.
- ———. 2010. Principles for Enhancing Corporate Governance. Basel: Bank for International Settlements.
- B. Chan and S. Cheung. 2004. *Corporate Governance in Asia*. Asia-Pacific Development Journal 11 (2).
- G. Capannelli and J. Menon. 2010. Dealing with Multiple Currencies in Transitional Economies: the scope for regional cooperation in Cambodia, the Lao People's Democratic Republic, and Viet Nam. Manila: Asian Development Bank.
- D. Cho and C. Rhee. 2013. Effects of Quantitative Easing on Asia: Capital Flows and Financial Markets. *ADB Economics Working Paper Series*. 350. Manila: Asian Development Bank.
- S. Claessens and B. Yurtoglu. 2012. Global Corporate Governance Forum Focus 10: Corporate Governance and Development—An Update. Washington: International Finance Corporation.
- Committee on International Economic Policy and Reform. 2012. Banks and Cross-Border Capital Flows: Policy Challenges and Regulatory Responses. Washington: Brookings Institution.
- E. Davis. 1999. Financial Data Needs for Macroprudential Surveillance—What are the Key Indicators for Risks to Domestic Financial Stability? *Handbooks in Central Banking Lecture Series*. 2. London: Bank of England.
- G. Estrada, D. Park, and A. Ramayandi. 2010. Financial Development and Economic Growth in Developing Asia. *ADB Economics Working Paper Series*. 233. Manila: Asian Development Bank.

- 40 | Working Paper Series on Regional Economic Integration No. 134
- O. Evans et al. 2000. Macroprudential Indicators of Financial System Soundness. *IMF* Occasional Paper. 192. Washington, DC: IMF.
- Financial Stability Board, IMF, and BIS. 2011. *Macroprudential Policy Tools and Frameworks*. Update to G20 Finance Ministers and Central Bank Governors. Basel.
- R. Gelos and S-J Wei. 2002. Transparency and International Investor Behavior. *NBER Working Paper Series*. 9260. Cambridge, MA: National Bureau of Economic Research.
- Government of Brunei Darussalam. Brunei Economic Development Board. Brunei's National Vision. http://www.bedb.com.bn/why_wawasan2035.html (accessed 3 June 2013).
- J-H Hahm, H. Shin, and K. Shin. 2012. Noncore Bank Liabilities and Financial Vulnerability. NBER Working Paper. 18428. Cambridge, MA: National Bureau of Economic Research.
- L. Hoogduin, ed. 2010. Macroprudential Instruments and Frameworks: a stocktaking of issues and experiences. Report submitted by the Committee on the Global Financial System. *CGFS Papers*. 38. Basel: Bank for International Settlements.
- IMF. 2001. IMF Survey Supplement. 30. Washington, DC: International Monetary Fund. pp. 7–8.
- M. Kawai and M. Lamberte (eds.). 2010. *Managing Capital Flows: The Search for a Framework*. Cheltenham, UK: Asian Development Bank Institute and Edward Elgar.
- M. Kose et. al. 2003. Effects of Financial Globalization on Developing Countries: Some Empirical Evidence. *IMF Occasional Paper*. 220. Washington, DC: IMF.
- M. Liu and F. Moshirian. Forthcoming. The Evolving Global Regulatory Framework and its Impact on Asia. Manila: ADB.
- OECD. 2004. OECD Principles of Corporate Governance. Paris: Organisation for Economic Co-operation and Development.
- OECD. 2010. Corporate Governance and the Financial Crisis: conclusions and emerging good practices to enhance implementation of the principles. Paris: Organisation for Economic Co-operation and Development.
- H. Sharif and S. Zaidansyah. 2004. Corporate Governance in Asia. In *Global Corporate Governance Guide 2004: Best Practice in the Boardroom.* Manila: Asian Development Bank.
- Shin, Hyun Song and Kwanho Shin. 2011. Procyclicality and Monetary Aggregates. *NBER Working Paper*. 16836. Cambridge, MA: National Bureau of Economic Research.

ADB Working Paper Series on Regional Economic Integration^{*}

- 1. "The ASEAN Economic Community and the European Experience" by Michael G. Plummer
- 2. "Economic Integration in East Asia: Trends, Prospects, and a Possible Roadmap" by Pradumna B. Rana
- 3. "Central Asia after Fifteen Years of Transition: Growth, Regional Cooperation, and Policy Choices" by Malcolm Dowling and Ganeshan Wignaraja
- 4. "Global Imbalances and the Asian Economies: Implications for Regional Cooperation" by Barry Eichengreen
- 5. "Toward Win-Win Regionalism in Asia: Issues and Challenges in Forming Efficient Trade Agreements" by Michael G. Plummer
- 6. "Liberalizing Cross-Border Capital Flows: How Effective Are Institutional Arrangements against Crisis in Southeast Asia" by Alfred Steinherr, Alessandro Cisotta, Erik Klär, and Kenan Šehović
- 7. "Managing the Noodle Bowl: The Fragility of East Asian Regionalism" by Richard E. Baldwin
- 8. "Measuring Regional Market Integration in Developing Asia: A Dynamic Factor Error Correction Model (DF-ECM) Approach" by Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, and Pilipinas F. Quising
- 9. "The Post-Crisis Sequencing of Economic Integration in Asia: Trade as a Complement to a Monetary Future" by Michael G. Plummer and Ganeshan Wignaraja
- 10. "Trade Intensity and Business Cycle Synchronization: The Case of East Asia" by Pradumna B. Rana
- 11. "Inequality and Growth Revisited" by Robert J. Barro
- 12. "Securitization in East Asia" by Paul Lejot, Douglas Arner, and Lotte Schou-Zibell
- 13. "Patterns and Determinants of Cross-border Financial Asset Holdings in East Asia" by Jong-Wha Lee
- 14. "Regionalism as an Engine of Multilateralism: A Case for a Single East Asian FTA" by Masahiro Kawai and Ganeshan Wignaraja
- 15. "The Impact of Capital Inflows on Emerging East Asian Economies: Is Too Much Money Chasing Too Little Good?" by Soyoung Kim and Doo Yong Yang

- 42 | Working Paper Series on Regional Economic Integration No. 134
 - 16. "Emerging East Asian Banking Systems Ten Years after the 1997/98 Crisis" by Charles Adams
 - 17. "Real and Financial Integration in East Asia" by Soyoung Kim and Jong-Wha Lee
 - 18. "Global Financial Turmoil: Impact and Challenges for Asia's Financial Systems" by Jong-Wha Lee and Cyn-Young Park
 - 19. "Cambodia's Persistent Dollarization: Causes and Policy Options" by Jayant Menon
 - 20. "Welfare Implications of International Financial Integration" by Jong-Wha Lee and Kwanho Shin
 - 21. "Is the ASEAN-Korea Free Trade Area (AKFTA) an Optimal Free Trade Area?" by Donghyun Park, Innwon Park, and Gemma Esther B. Estrada
 - 22. "India's Bond Market—Developments and Challenges Ahead" by Stephen Wells and Lotte Schou- Zibell
 - 23. "Commodity Prices and Monetary Policy in Emerging East Asia" by Hsiao Chink Tang
 - 24. "Does Trade Integration Contribute to Peace?" by Jong-Wha Lee and Ju Hyun Pyun
 - 25. "Aging in Asia: Trends, Impacts, and Responses" by Jayant Menon and Anna Melendez-Nakamura
 - 26. "Re-considering Asian Financial Regionalism in the 1990s" by Shintaro Hamanaka
 - 27. "Managing Success in Viet Nam: Macroeconomic Consequences of Large Capital Inflows with Limited Policy Tools" by Jayant Menon
 - 28. "The Building Block versus Stumbling Block Debate of Regionalism: From the Perspective of Service Trade Liberalization in Asia" by Shintaro Hamanaka
 - 29. "East Asian and European Economic Integration: A Comparative Analysis" by Giovanni Capannelli and Carlo Filippini
 - 30. "Promoting Trade and Investment in India's Northeastern Region" by M. Govinda Rao
 - 31. "Emerging Asia: Decoupling or Recoupling" by Soyoung Kim, Jong-Wha Lee, and Cyn-Young Park

- 32. "India's Role in South Asia Trade and Investment Integration" by Rajiv Kumar and Manjeeta Singh
- 33. "Developing Indicators for Regional Economic Integration and Cooperation" by Giovanni Capannelli, Jong-Wha Lee, and Peter Petri
- 34. "Beyond the Crisis: Financial Regulatory Reform in Emerging Asia" by Chee Sung Lee and Cyn-Young Park
- 35. "Regional Economic Impacts of Cross-Border Infrastructure: A General Equilibrium Application to Thailand and Lao People's Democratic Republic" by Peter Warr, Jayant Menon, and Arief Anshory Yusuf
- 36. "Exchange Rate Regimes in the Asia-Pacific Region and the Global Financial Crisis" by Warwick J. McKibbin and Waranya Pim Chanthapun
- 37. "Roads for Asian Integration: Measuring ADB's Contribution to the Asian Highway Network" by Srinivasa Madhur, Ganeshan Wignaraja, and Peter Darjes
- 38. "The Financial Crisis and Money Markets in Emerging Asia" by Robert Rigg and Lotte Schou-Zibell
- 39. "Complements or Substitutes? Preferential and Multilateral Trade Liberalization at the Sectoral Level" by Mitsuyo Ando, Antoni Estevadeordal, and Christian Volpe Martincus
- 40. "Regulatory Reforms for Improving the Business Environment in Selected Asian Economies—How Monitoring and Comparative Benchmarking can Provide Incentive for Reform" by Lotte Schou-Zibell and Srinivasa Madhur
- 41. "Global Production Sharing, Trade Patterns, and Determinants of Trade Flows in East Asia" by Prema-chandra Athukorala and Jayant Menon
- 42. "Regionalism Cycle in Asia (-Pacific): A Game Theory Approach to the Rise and Fall of Asian Regional Institutions" by Shintaro Hamanaka
- 43. "A Macroprudential Framework for Monitoring and Examining Financial Soundness" by Lotte Schou-Zibell, Jose Ramon Albert, and Lei Lei Song
- 44. "A Macroprudential Framework for the Early Detection of Banking Problems in Emerging Economies" by Claudio Loser, Miguel Kiguel, and David Mermelstein
- 45. "The 2008 Financial Crisis and Potential Output in Asia: Impact and Policy Implications" by Cyn-Young Park, Ruperto Majuca, and Josef Yap
- 46. "Do Hub-and-Spoke Free Trade Agreements Increase Trade? A Panel Data Analysis" by Jung Hur, Joseph Alba, and Donghyun Park

- 44 | Working Paper Series on Regional Economic Integration No. 134
 - 47. "Does a Leapfrogging Growth Strategy Raise Growth Rate? Some International Evidence" by Zhi Wang, Shang-Jin Wei, and Anna Wong
 - 48. "Crises in Asia: Recovery and Policy Responses" by Kiseok Hong and Hsiao Chink Tang
 - 49. "A New Multi-Dimensional Framework for Analyzing Regional Integration: Regional Integration Evaluation (RIE) Methodology" by Donghyun Park and Mario Arturo Ruiz Estrada
 - 50. "Regional Surveillance for East Asia: How Can It Be Designed to Complement Global Surveillance?" by Shinji Takagi
 - 51. "Poverty Impacts of Government Expenditure from Natural Resource Revenues" by Peter Warr, Jayant Menon, and Arief Anshory Yusuf
 - 52. "Methods for Ex Ante Economic Evaluation of Free Trade Agreements" by David Cheong
 - 53. "The Role of Membership Rules in Regional Organizations" by Judith Kelley
 - 54. "The Political Economy of Regional Cooperation in South Asia" by V.V. Desai
 - 55. "Trade Facilitation Measures under Free Trade Agreements: Are They Discriminatory against Non-Members?" by Shintaro Hamanaka, Aiken Tafgar, and Dorothea Lazaro
 - 56. "Production Networks and Trade Patterns in East Asia: Regionalization or Globalization?" by Prema-chandra Athukorala
 - 57. "Global Financial Regulatory Reforms: Implications for Developing Asia" by Douglas W. Arner and Cyn-Young Park
 - 58. "Asia's Contribution to Global Rebalancing" by Charles Adams, Hoe Yun Jeong, and Cyn-Young Park
 - 59. "Methods for Ex Post Economic Evaluation of Free Trade Agreements" by David Cheong
 - 60. "Responding to the Global Financial and Economic Crisis: Meeting the Challenges in Asia" by Douglas W. Arner and Lotte Schou-Zibell
 - 61. "Shaping New Regionalism in the Pacific Islands: Back to the Future?" by Satish Chand
 - 62. "Organizing the Wider East Asia Region" by Christopher M. Dent
 - 63. "Labour and Grassroots Civic Interests In Regional Institutions" by Helen E.S. Nesadurai

- 64. "Institutional Design of Regional Integration: Balancing Delegation and Representation" by Simon Hix
- 65. "Regional Judicial Institutions and Economic Cooperation: Lessons for Asia?" by Erik Voeten
- 66. "The Awakening Chinese Economy: Macro and Terms of Trade Impacts on 10 Major Asia-Pacific Countries" by Yin Hua Mai, Philip Adams, Peter Dixon, and Jayant Menon
- 67. "Institutional Parameters of a Region-Wide Economic Agreement in Asia: Examination of Trans-Pacific Partnership and ASEAN+α Free Trade Agreement Approaches" by Shintaro Hamanaka
- 68. "Evolving Asian Power Balances and Alternate Conceptions for Building Regional Institutions" by Yong Wang
- 69. "ASEAN Economic Integration: Features, Fulfillments, Failures, and the Future" by Hal Hill and Jayant Menon
- 70. "Changing Impact of Fiscal Policy on Selected ASEAN Countries" by Hsiao Chink Tang, Philip Liu, and Eddie C. Cheung
- 71. "The Organizational Architecture of the Asia-Pacific: Insights from the New Institutionalism" by Stephan Haggard
- 72. "The Impact of Monetary Policy on Financial Markets in Small Open Economies: More or Less Effective During the Global Financial Crisis?" by Steven Pennings, Arief Ramayandi, and Hsiao Chink Tang
- 73. "What do Asian Countries Want the Seat at the High Table for? G20 as a New Global Economic Governance Forum and the Role of Asia" by Yoon Je Cho
- 74. "Asia's Strategic Participation in the Group of 20 for Global Economic Governance Reform: From the Perspective of International Trade" by Taeho Bark and Moonsung Kang
- 75. "ASEAN's Free Trade Agreements with the People's Republic of China, Japan, and the Republic of Korea: A Qualitative and Quantitative Analysis" by Gemma Estrada, Donghyun Park, Innwon Park, and Soonchan Park
- 76. "ASEAN-5 Macroeconomic Forecasting Using a GVAR Model" by Fei Han and Thiam Hee Ng
- 77. "Early Warning Systems in the Republic of Korea: Experiences, Lessons, and Future Steps" by Hyungmin Jung and Hoe Yun Jeong

- 46 | Working Paper Series on Regional Economic Integration No. 134
 - 78. "Trade and Investment in the Greater Mekong Subregion: Remaining Challenges and the Unfinished Policy Agenda" by Jayant Menon and Anna Cassandra Melendez
 - 79. "Financial Integration in Emerging Asia: Challenges and Prospects" by Cyn-Young Park and Jong-Wha Lee
 - 80. "Sequencing Regionalism: Theory, European Practice, and Lessons for Asia" by Richard E. Baldwin
 - 81. "Economic Crises and Institutions for Regional Economic Cooperation" by C. Randall Henning
 - 82. "Asian Regional Institutions and the Possibilities for Socializing the Behavior of States" by Amitav Acharya
 - 83. "The People's Republic of China and India: Commercial Policies in the Giants" by Ganeshan Wignaraja
 - 84. "What Drives Different Types of Capital Flows and Their Volatilities?" by Rogelio Mercado and Cyn-Young Park
 - 85. "Institution Building for African Regionalism" by Gilbert M. Khadiagala
 - 86. "Impediments to Growth of the Garment and Food Industries in Cambodia: Exploring Potential Benefits of the ASEAN-PRC FTA" by Vannarith Chheang and Shintaro Hamanaka
 - 87. "The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation" by Hyun-Hoon Lee, Donghyun Park, and Jing Wang
 - 88. "Utilizing the Multiple Mirror Technique to Assess the Quality of Cambodian Trade Statistics" by Shintaro Hamanaka
 - 89. "Is Technical Assistance under Free Trade Agreements WTO-Plus?" A Review of Japan-ASEAN Economic Partnership Agreements" by Shintaro Hamanaka
 - 90. "Intra-Asia Exchange Rate Volatility and Intra-Asia Trade: Evidence by Type of Goods" by Hsiao Chink Tang
 - 91. "Is Trade in Asia Really Integrating?" by Shintaro Hamanaka
 - 92. "The PRC's Free Trade Agreements with ASEAN, Japan, and the Republic of Korea: A Comparative Analysis" by Gemma Estrada, Donghyun Park, Innwon Park, and Soonchan Park

- 93. "Assessing the Resilience of ASEAN Banking Systems: The Case of the Philippines" by Jose Ramon Albert and Thiam Hee Ng
- 94. "Strengthening the Financial System and Mobilizing Savings to Support More Balanced Growth in ASEAN+3" by A. Noy Siackhachanh
- 95. "Measuring Commodity-Level Trade Costs in Asia: The Basis for Effective Trade Facilitation Policies in the Region" by Shintaro Hamanaka and Romana Domingo
- 96. "Why do Imports Fall More than Exports Especially During Crises? Evidence from Selected Asian Economies" by Hsiao Chink Tang
- 97. "Determinants of Local Currency Bonds and Foreign Holdings: Implications for Bond Market Development in the People's Republic of China" by Kee-Hong Bae
- 98. "ASEAN-China Free Trade Area and the Competitiveness of Local Industries: A Case Study of Major Industries in the Lao People's Democratic Republic" by Leebeer Leebouapao, Sthabandith Insisienmay, and Vanthana Nolintha
- 99. "The Impact of ACFTA on People's Republic of China-ASEAN Trade: Estimates Based on an Extended Gravity Model for Component Trade" by Yu Sheng, Hsiao Chink Tang, and Xinpeng Xu
- 100. "Narrowing the Development Divide in ASEAN: The Role of Policy" by Jayant Menon
- 101. "Different Types of Firms, Products, and Directions of Trade: The Case of the People's Republic of China" by Hyun-Hoon Lee, Donghyun Park, and Jing Wang
- 102. "Anatomy of South–South FTAs in Asia: Comparisons with Africa, Latin America, and the Pacific Islands" by Shintaro Hamanaka
- 103. "Japan's Education Services Imports: Branch Campus or Subsidiary Campus?" by Shintaro Hamanaka
- 104. "A New Regime of SME Finance in Emerging Asia: Empowering Growth-Oriented SMEs to Build Resilient National Economies" by Shigehiro Shinozaki
- 105. "Critical Review of East Asia South America Trade " by Shintaro Hamanaka and Aiken Tafgar
- 106. "The Threat of Financial Contagion to Emerging Asia's Local Bond Markets: Spillovers from Global Crises" by Iwan J. Azis, Sabyasachi Mitra, Anthony Baluga, and Roselle Dime

- 48 | Working Paper Series on Regional Economic Integration No. 134
 - 107. "Hot Money Flows, Commodity Price Cycles, and Financial Repression in the US and the People's Republic of China: The Consequences of Near Zero US Interest Rates" by Ronald McKinnon and Zhao Liu
 - 108. "Cross-Regional Comparison of Trade Integration: The Case of Services" by Shintaro Hamanaka
 - 109. "Preferential and Non-Preferential Approaches to Trade Liberalization in East Asia: What Differences Do Utilization Rates and Reciprocity Make?" by Jayant Menon
 - 110. "Can Global Value Chains Effectively Serve Regional Economic Development in Asia?" by Hans-Peter Brunner
 - 111. "Exporting and Innovation: Theory and Firm-Level Evidence from the People's Republic of China" by Faqin Lin and Hsiao Chink Tang
 - 112. "Supporting the Growth and Spread of International Production Networks in Asia: How Can Trade Policy Help?" by Jayant Menon
 - 113. "On the Use of FTAs: A Review of Research Methodologies" by Shintaro Hamanaka
 - 114. "The People's Republic of China's Financial Policy and Regional Cooperation in the Midst of Global Headwinds" by Iwan J. Azis
 - 115. "The Role of International Trade in Employment Growth in Micro- and Small Enterprises: Evidence from Developing Asia" by Jens Krüger
 - 116. "Impact of Euro Zone Financial Shocks on Southeast Asian Economies" by Jayant Menon and Thiam Hee Ng
 - 117. "What is Economic Corridor Development and What Can It Achieve in Asia's Subregions?" by Hans-Peter Brunner
 - 118. "The Financial Role of East Asian Economies in Global Imbalances: An Econometric Assessment of Developments after the Global Financial Crisis" by Hyun-Hoon Lee and Donghyun Park
 - 119. "Learning by Exporting: Evidence from India" by Apoorva Gupta, Ila Patnaik, and Ajay Shah
 - 120. "FDI Technology Spillovers and Spatial Diffusion in the People's Republic of China" by Mi Lin and Yum K. Kwan
 - 121. "Capital Market Financing for SMEs: A Growing Need in Emerging Asia" by Shigehiro Shinozaki

- 122. "Terms of Trade, Foreign Direct Investment, and Development: A Case of Intra-Asian Kicking Away the Ladder?" by Konstantin M. Wacker, Philipp Grosskurth, and Tabea Lakemann
- 123. "Can Low Interest Rates be Harmful: An Assessment of the Bank Risk-Taking Channel in Asia" by Arief Ramayandi, Umang Rawat, and Hsiao Chink Tang
- 124. "Explaining Foreign Holdings of Asia's Debt Securities" by Charles Yuji Horioka, Takaaki Nomoto, and Akiko Terada-Hagiwara
- 125. "South Caucasus–People's Republic of China Bilateral Free Trade Agreements: Why It Matters" by Hasmik Hovhanesian and Heghine Manasyan"
- 126. "Enlargement of Economic Framework in Southeast Asia and Trade Flows in Lao PDR" by Sithanonxay Suvannaphakdy, Hsiao Chink Tang, and Alisa DiCaprio
- 127. "The End of Grand Expectations: Monetary and Financial Integration After the Crisis in Europe" by Heribert Dieter
- 128. "The Investment Version of the Asian Noodle Bowl: The Proliferation of International Investment Agreements (IIAs)" by Julien Chaisse and Shintaro Hamanaka
- 129. "Why Do Countries Enter into Preferential Agreements on Trade in Services?: Assessing the Potential for Negotiated Regulatory Convergence in Asian Services Markets" by Pierre Sauvé and Anirudh Shingal
- 130. "Analysis of Informal Obstacles to Cross-Border Economic Activity between Kazakhstan and Uzbekistan" by Roman Vakulchuk and Farrrukh Irnazarov
- 131. "The Nexus between Anti-Dumping Petitions and Exports during the Global Financial Crisis: Evidence on the People's Republic of China" by Faqin Lin, Hsiao Chink Tang, and Lin Wang
- 132. "Study of Non-Notified Trade Agreements to WTO: The Case of Asia-Pacific" by Shintaro Hamanaka
- 133. "Equity Home Bias Financial Integration and Regulatory Reforms" by Cyn-Young Park and Rogelio V. Mercado, Jr.

*These papers can be downloaded from (ARIC) http://aric.adb.org/archives.php? section=0&subsection=workingpapers or (ADB) http://www.adb.org/publications/series/ regional-economic-integration-working-papers

Financial Monitoring in the New ASEAN-5 Countries

The five newest members of the Association of Southeast Asian Nations (ASEAN)—Brunei Darussalam, Cambodia, the Lao People's Democratic Republic, Myanmar, and Viet Nam—continue to develop their financial sectors and catch up with the rest of southeast Asia prior to the launch of ASEAN Economic Community in 2015. This paper examines the status and concerns regarding the five nation's financial sector development and the impact of global financial reforms after the crisis.

About the Asian Development Bank

ADB's vision is an Asia and Pacific region free of poverty. Its mission is to help its developing member countries reduce poverty and improve the quality of life of their people. Despite the region's many successes, it remains home to approximately two-thirds of the world's poor: 1.6 billion people who live on less than \$2 a day, with 733 million struggling on less than \$1.25 a day. ADB is committed to reducing poverty through inclusive economic growth, environmentally sustainable growth, and regional integration.

Based in Manila, ADB is owned by 67 members, including 48 from the region. Its main instruments for helping its developing member countries are policy dialogue, loans, equity investments, guarantees, grants, and technical assistance.



ASIAN DEVELOPMENT BANK

6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org Publication Stock No. WPS146550